

IntechOpen

Human Capital and Competences in Project Management

Edited by Manuel Otero-Mateo and Andres Pastor-Fernandez





HUMAN CAPITAL AND COMPETENCES IN PROJECT MANAGEMENT

Edited by Manuel Otero-Mateo and Andres Pastor-Fernandez

Human Capital and Competences in Project Management

http://dx.doi.org/10.5772/67268 Edited by Manuel Otero-Mateo and Andres Pastor-Fernandez

Contributors

Anabela Alves, Celina Leão, Francisco Moreira, Senhorinha Teixeira, Alberto Cerezo-Narváez, María José Bastante-Ceca, José Luis Yagüe Blanco, Alexandr Yuriy Kokovikhin, Makoto Katoh, Toshio Haga, Muneyoshi Iyota, Keiko Natori, Yutaka Kawata, Hiroyuki Kobayashi, Kazuo Kumamoto, Tsutomu Yoshimura Tsutomu Yoshimura, Mara Mataveli, Alfonso J. Gil, Manuel Otero-Mateo, Andres Pastor-Fernandez

© The Editor(s) and the Author(s) 2018

The moral rights of the and the author(s) have been asserted.

All rights to the book as a whole are reserved by INTECH. The book as a whole (compilation) cannot be reproduced, distributed or used for commercial or non-commercial purposes without INTECH's written permission. Enquiries concerning the use of the book should be directed to INTECH rights and permissions department (permissions@intechopen.com).

Violations are liable to prosecution under the governing Copyright Law.

CC BY

Individual chapters of this publication are distributed under the terms of the Creative Commons Attribution 3.0 Unported License which permits commercial use, distribution and reproduction of the individual chapters, provided the original author(s) and source publication are appropriately acknowledged. If so indicated, certain images may not be included under the Creative Commons license. In such cases users will need to obtain permission from the license holder to reproduce the material. More details and guidelines concerning content reuse and adaptation can be foundat http://www.intechopen.com/copyright-policy.html.

Notice

Statements and opinions expressed in the chapters are these of the individual contributors and not necessarily those of the editors or publisher. No responsibility is accepted for the accuracy of information contained in the published chapters. The publisher assumes no responsibility for any damage or injury to persons or property arising out of the use of any materials, instructions, methods or ideas contained in the book.

First published in Croatia, 2018 by INTECH d.o.o. eBook (PDF) Published by IN TECH d.o.o. Place and year of publication of eBook (PDF): Rijeka, 2019. IntechOpen is the global imprint of IN TECH d.o.o. Printed in Croatia

Legal deposit, Croatia: National and University Library in Zagreb

Additional hard and PDF copies can be obtained from orders@intechopen.com

Human Capital and Competences in Project Management Edited by Manuel Otero-Mateo and Andres Pastor-Fernandez p. cm. Print ISBN 978-953-51-3786-3 Online ISBN 978-953-51-3787-0 eBook (PDF) ISBN 978-953-51-4018-4

We are IntechOpen, the first native scientific publisher of Open Access books

<u>3.300</u>+ Open access books available <u>107,000+</u> 113M+

International authors and editors

Downloads

15Countries delivered to Our authors are among the

Top 1% most cited scientists

12.2%

Contributors from top 500 universities



WEB OF SCIENCE

Selection of our books indexed in the Book Citation Index in Web of Science[™] Core Collection (BKCI)

Interested in publishing with us? Contact book.department@intechopen.com

Numbers displayed above are based on latest data collected. For more information visit www.intechopen.com



Meet the editor



Manuel Otero-Mateo is a doctor in Engineering and Architecture from the University of Cádiz (UCA), Spain. He received his DEA program interuniversity PhD degree in Manufacturing Engineering from the University of Malaga, Spain. Dr. Otero-Mateo is an engineer of industrial organization and industrial engineering (especially in Industrial Electronics) at UCA, Spain. He

is a certified project manager, IPMA Level C. His expertise is in PLC Siemens S7-300 and S7-400 with 10 years of experience in Management and Project Management in the Agrifood sector. Dr. Otero-Mateo is a professor at the Department of Design Engineering of the University of Seville, and currently, he is an assistant professor at the Department of Mechanical Engineering and Industrial Design of UCA, Spain. He is a member of the Editorial Committee of the Publication Services at UCA and attached to TEP955 Research Group—Engineering and Technology for Occupational Health and Safety (INTELPREV).



Andres Pastor Fernandez obtained his PhD degree in Engineering from the University of Cádiz (UCA), Spain. He is an engineer of industrial organization and industrial technical engineer (Mechanical) at UCA, Spain. He is a certified project manager, IPMA Level C. He is a project manager for the petrochemical sector, besides holding management positions in the naval industry

(IZAR S.A.), with more than 10 years of experience in Project Management and Management. He is an associate professor and a director at the Department of Mechanical Engineering and Industrial Design of UCA, Spain. He is a principal investigator of TEP955 Research Group—Engineering and Technology for Occupational Health and Safety (INTELPREV).

Contents

Preface XI

- Section 1 Introduction 1
- Chapter 1 Introductory Chapter: Human Capital, Knowledge Management and Competences in Project Management 3 Manuel Otero-Mateo and Andres Pastor-Fernandez
- Section 2 Education Competences 7
- Chapter 2 Project-Based Learning and its Effects on Freshmen Social Skills in an Engineering Program 9 Anabela C. Alves, Celina P. Leão, Francisco Moreira and Senhorinha Teixeira
- Chapter 3 Competences in Project Management: A Case Study in Osaka Institute of Technology 27 Makoto Katoh, Yutaka Kawata, Toshio Haga, Hiroyuki Kobayashi, Tsutomu Yoshimura, Kazuo Kumamoto, Muneyoshi Iyota and Keiko Natori
- Section 3 Labour Market Competences 43
- Chapter 4 **Project Management and Learning: The Learning Project 45** Alfonso J. Gil and Mara Mataveli
- Chapter 5 Culture and Values Competence in International Investment Projects: HR Management Approach 71 Alexandr Kokovikhin

Section 4 Key Competences for Employment 85

Chapter 6 Traceability of Intra- and Interpersonal Skills: From Education to Labor Market 87

Alberto Cerezo-Narváez, María José Bastante Ceca and José Luis Yagüe Blanco

Preface

The acquisition of professional skills and the development of human capital are two of the great challenges facing the business environment today. With the latest regulatory changes (ISO 9001 and ISO 14001 published in 2015), the assurance of the skills of workers acquires a special relevance, and it is therefore necessary to reflect on the acquisition of skills of future workers who acquire these in the university environment as well as their subsequent learning in the workplace.

The book tries to provide a framework for reflection on the relationships between the university education environment and the labor market, we are not only talking about that, by having a qualification, the worker is guaranteed the appropriate skills to develop his work, but his competencies must be maintained over time and improved with continuous learning throughout the working life. For this, the book is structured in four sections.

The first section Introduction – short introduction to topic on Human Capital, Knowledge Management and Competences in Project Management.

The second section, corresponding to the competences in the educational field, analyzes projectbased learning (PBL) and its effects on social skills in engineering studies, as well as the development of skills in Project Management.

The third section analyzes the competences of the labor market, the discipline of Project Management, and the development of professional competences proposed by the International Project Management Association through its IPMA Competence Baseline (ICB 4.0) standard and its relationship with the management of Human Resources in business organizations.

The fourth and last section analyzes the key competences of employment, being necessary to an alignment between the university and the business environment. This study completes the book and provides the traceability of intra- and interpersonal skills, besides being a support to the professional competences in Management of Projects and the ICB 4.0 standard.

The research line started by this book allows us to glimpse the importance of a greater communication between the educational field and the working world. This connection should also foster greater development of interpersonal skills, acting as a facilitator for the adaptation of the worker to the changes that the labor world is experiencing in the advent of Industry 4.0. Finally, generating knowledge and strengthening human capital is vital, because without people, the challenge of the 4th industrial revolution will not be possible.

Finally, we want to thank all the authors who participated in this book, as they provide different points of view to the challenges facing the education and the world of work to give a better service to society.

> Manuel Otero-Mateo and Andres Pastor-Fernandez University of Cadiz, Department of Mechanical Engineering and Industrial Design, Puerto Real (Cadiz), Spain

Section 1

Introduction

Introductory Chapter: Human Capital, Knowledge Management and Competences in Project Management

Manuel Otero-Mateo and Andres Pastor-Fernandez

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.73639

1. Introduction

Although there are several studies that have researched into the effects behavioral competences have on management and knowledge acquisition, their actual benefits have not been proven yet. Therefore, the aim of this research book is to shed light on such benefits.

The people competence has a strong influence on the strategy of human resources management, affecting daily aspects, thought patterns, and behavioral modes of executive management and its employees. Only when the culture, the organizations, and the strategy of human resources management complement each other, the strategies are effective and can favor the creation of competitive advantages. From a business perspective, there is a strong relationship between human capital and success. The relationship becomes stronger when the tasks to be performed are complex and require highly qualified employees. For an integral development of the human factor in all its dimensions, both personal and professional, social competences must be an important factor.

This challenge is facing the industries currently with the fourth industrial revolution, the socalled Industry 4.0, in which the worker's role moves from a more physical level to a higher level of organization and management (supervisor) of the business system, where interpersonal competences, social skills and of course the knowledge of human capital acquire a new dimension.

To help achieve this business excellence, we propose to use the methodological framework in Project Management and Management, specifically the IPMA Individual Competence Baseline [1]. This model is based on competencies, defining these as the application of Knowledge (collection of information and experience that an individual possesses), Skills (specific technical capabilities that enable an individual to perform a task) and Ability (effective delivery of knowledge and skills in a given context).



© 2018 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. This book intends to provide the reader with a comprehensive overview of the current stateof-the-art in human capital, competences in Project Management, learning strategies and their influence on organizations.

2. Competencies in project management as support for knowledge management

In today's globalized world, knowledge management has become an essential tool for achieving economic growth, corporate development, and competitiveness. Knowledge management must also involve a balance between good practices and productive processes [2]. Therefore, acquisition of knowledge (generation of ideas and opportunities) as well as its implementation in processes, where it can be put into practice, is of great importance [3].

Knowledge can be defined as the combination of experience, know-how, values, information, perception and ideas that create a framework of mind that helps people to assess and generate new ideas, knowledge and experience [4]. Human capital (HC) plays a key role in knowledge since it adds value to the knowledge and competences of the organization's people, and the capacity to generate them is useful to achieve the organization's mission [5, 6].

Generally speaking, HC refers to the knowledge acquired by people increasing their productivity and adding value to his contribution. HC includes the employee's personal contacts and relationships, as well as other individual qualities such as reputation, loyalty, multitasking or flexibility.

Drawing on the International Project Management Association model on Project Management, the study examines different behavioral competence elements. With a project management approach, the book draws on the theoretical notions as well as the professional experience of the stakeholders and mainly on trainers.

Despite being related to the management of knowledge in project management processes and thus to human capital, technical and contextual competencies are somewhat independent of an individual's attitudes and skills in the performance of his or her tasks. While it is indeed necessary to possess knowledge, it is also necessary to know how to transfer it to create value in the organization, and it is here that behavioral competencies play a fundamental role in the value creation chain.

The book presents a modest vision about project-based training and learning and the competence demands of organizations to end with a chapter on the traceability of intra- and interpersonal competences between the training field and the labor market using the guide as a support tool IPMA ICB[®] [1].

Author details

Manuel Otero-Mateo* and Andres Pastor-Fernandez

*Address all correspondence to: manuel.otero@uca.es

Department of Mechanical Engineering and Industrial Design, University of Cadiz, Puerto Real Cadiz, Spain

References

- [1] International Project Management Association. Individual Competence Baseline for Project, Programme & Portfolio Management. 4th ed. Zurich: IPMA; 2015
- [2] Brown J, Duguid P. Balancing act: How to capture knowledge without killing it. Harvard Business Review. 2000;**78**(3):73-80
- [3] Coetzee JC, van Beek WSB, Buys A. A practical knowledge management framework within the pyrometallurgical industry. Journal of the Southern African Institute of Mining and Metallurgy 2012;**112**(7):621-630
- [4] European Committee for Standardization (CEN). European Guide to good Practice in Knowledge Management. Part 1 to 5. CEN: Brussels; 2004
- [5] Hatch N, Dyer J. Human capital and learning as a source of sustainable competitive advantage. Strategic Management Journal. 2004;25(12):1155-1178. DOI: 10.1002/smj.421
- [6] Palacios-Marques D, Gil-Pechuan I, Lim S. Improving human capital through knowledge management practices in knowledge-intensive business services. Service Business. 2011;5(2):99-112. DOI: 10.1007/s11628-011-0104-z

Education Competences

Project-Based Learning and its Effects on Freshmen Social Skills in an Engineering Program

Anabela C. Alves, Celina P. Leão, Francisco Moreira and Senhorinha Teixeira

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.72054

Abstract

This chapter reports some effects of project-based learning (PBL) on development of social skills on Industrial Engineering freshmen (first-year) students. PBL is an active learning and student-centered methodology that promotes skills development such as the 4C—Critical thinking, Communication, Collaboration and Creativity. These skills should be an integral part of the expected engineering competences needed for professional proficiency, in order to comply with a changeable world and instable marketplaces, which require competences well beyond the technical ones. Through a survey to the first-year students and interviews to recently graduated professionals, some interesting results about the effects of PBL on social skills development were acquired. Some of these results relate to the recognition of acquiring competences during the project development and the usefulness and applicability of these in their daily professional activity. Among others, these encompass the capability to lead projects and produce effective work within multidisciplinary teams, to deal with conflicts, and to provide effective oral and written communication and capability to adapt to different work environments and assuming responsibilities, reflecting and assessing the own learning and the work of others, and to respect the attitudes and work of others.

Keywords: project-based learning, soft skills, engineering education, active learning methodologies, competencies development

1. Introduction

The Fourth Industrial Revolution, promoted by, what is known in the occidental world as Industry4.0, will put some challenges to the new engineers, as alerted by the report of the National Academy of Science and Engineering [1]. It is envisioned that the way work will be



organized in the future will enable the release of workers doing routine tasks, appealing to their skills for more creative and value-added activities. Additionally, they will be called to develop more complex products and systems and to manage them efficiently through new methods, tools and technologies [e.g., by using, among others, augmented reality (AR), virtual reality (VR), cyber-physical systems (CPS)] and to use transdisciplinary perspectives [2].

Work organization and design changes will imply a total new role of workers, increasing their responsibility and enhancing their personal development. According to the report of the American Management Association [3], the skills to deal with the fast pace of change in businesses are beyond the traditional "Three Rs" of reading, writing and arithmetic to a new set of skills, the "Four Cs": Critical thinking, Communication, Collaboration and Creativity, to enable workers to think critically, solve problems, innovate, collaborate and communicate more effectively.

The changes described require a new thinking about the way prospective workers are trained. This implies modifications in teaching institutions and pedagogical approaches. These modifications must educate the future workers to have more initiative, to possess excellent communication skills and the ability to organize their own work, as recommended by the reports referred above [1, 3] and others [4]. These needed skills fit in the set of the called Social Skills that according to the Business dictionary is the "Ability to communicate, persuade, and interact with other members of the society, without undue conflict or disharmony" [5]. Other definitions found are aligned with the one presented above, such as the definition of the Collins dictionary that states: "the skills that are necessary in order to communicate and interact with others" [6].

Active learning methodologies are particularly advocated, such as project-based learning (PBL), as methodologies capable of providing such skills [7–9]. Project-based learning is an active learning methodology that engages the students on their own learning, and puts them at the heart of competence development, including technical and transversal ones, like social skills. Indeed, PBL requires that team members ultimately bond with their peers in a student-centered approach, aiming at the development of technical proficiency in a number of subjects, for proposing a meaningful solution for an open-ended challenge. This prolific process requires that team members abandon their passive attitude toward learning and allows them to systematically exercise a number of distinct settings, which enrich their learning experiences. These are key in engineering programs where students are ought to develop competences for future practice, such as solving real-life problems, making convincing arguments (oral and written), leading teams, managing conflicts, working effectively within teams, considering social and environmental issues, interacting with others (e.g., colleagues, partners, clients) and being proactive and innovative.

This chapter discusses the PBL implemented in the Industrial Engineering and Management (IEM) program of first year since 2004–2005 as an effective tool to promote the defined social skills in the freshman. The instruments used to evidence this were the PBL process assessment survey applied in the end of semester to the first-year students and some interviews with the graduates PBL participants.

This chapter is divided into six sections. After the introduction, a brief literature review about PBL and skills is presented in Section 2. The research methodology is introduced in Section 3. Section 4 presents the context study describing the most important aspects of PBL. Section 5 points out the results and main findings and, finally, last section outlines some conclusions.

2. Literature review

According to Schmier [10], teaching is a sacred trust, a heavy responsibility, a privilege not to be taken lightly, a noble mission, a profound opportunity and making in life. This could not be more true when it is taken seriously, and in order to happen, it is necessary to be prepared to teach. This means recognizance and having present powerful ideas in teaching, as introduced by Gibbs and Habeshaw [11]: (1) students construct knowledge; (2) students need to see the whole picture; (3) students are selectively negligent; (4) students are driven by assessment; (5) students often only memorize; (6) students' attention is limited; (7) students can easily be overburdened; (8) adults learn differently; (9) students learn well by doing; (10) students learn well when they take responsibility for their learning and, finally, (11) students have feelings.

How to put in practice such ideas and implement teaching in effective learning? Active learning seems to be the solution. Active learning is defined by Bonwell and Eison [7] as instructional activities involving students in doing things and thinking about what they are doing. These activities are capable of creating excitement in the classroom at such a point that learning would be natural. Active learning activities should be capable to provide six levels of Bloom taxonomy: (1) knowledge, (2) comprehension, (3) application, (4) analysis, (5) synthesis and (6) evaluation and this same, revised [12]. Another taxonomy to be covered by active learning methodologies is Fink's taxonomy levels [13] of significant learning. Fink's view levels are as follows: foundational knowledge, application, integration, human dimensions, caring and learning how to learn. These authors referred that Bloom six-level taxonomy is no longer enough to the transformational learning practice.

Comparing these taxonomies, all are important in education, but, particularly, the last three from Fink' taxonomy, are, maybe, the most fundamental levels in an engineer training. In addition to these, engineer training demanded an holistic development of knowledge (head), dispositions (heart) and application (hands) and competency development of sustainability, systems and ethics provided by a Lean Engineering Education, as advocated by Flumerfelt et al. [14]. Such themes are not easy to teach/learn demanding active learning but also collaborative and cooperative approaches where the students become the center of their own learning. According to Prince [15], collaborative approaches can refer to any instructional method in which students work together in small groups toward a common goal, and cooperative as a structured form of workgroup where students pursue common goals while being assessed individually.

Project-based learning is an active learning methodology that involves collaborative and cooperative approaches. This could have different roots and different moments of interest and renaissance [16]. Nevertheless, it seems that the history of project method was systematically attached to the works of Dewey [17] and Kilpatrick [18, 19]. Both authors agree on the suitability of this method to prepare autonomous, independent and responsible citizens for their active practice social and democratic modes of behavior. These values are learned by the students while constructing the path to their own knowledge, combining theory with practice in a meaningful environment and a purposeful education. Some universities and colleges, such as the Aalborg University in Denmark, are converted to the project method adopting an approach of project (project-based learning) that could go from a single project in a course, the Task project, to a problem project [9].

In an engineering environment, some authors, for example, Powel and Weenk [20], named PBL as Project-Led Engineering Education (PLEE) and defined it as "Project-led engineering education focuses on team-based student activity relating to learning and to solving large-scale open-ended projects. Each project is usually supported by several theory-based lecture courses linked by a theme that labels the curriculum unit. A team of students tackles the project, provides a solution, and delivers by an agreed delivery time (a deadline) a 'team product', such as a prototype and a team report. Students show what they have learned by discussing with staff the 'team product' and reflecting on how they achieved it."

By doing such projects, students develop technical competences and transversal (or transferable) competences. Among these are the social skills, referred above, that in essence are related to the ability, as individual, to engage effectively with others. It is this that defines a person in terms of how he/she establishes healthy relations with others. When this happens, there are conditions for the knowledge sharing and growth. An intelligent individual only is recognizable as such if he/she has its recognition by the peers. So, social skills are utmost the trigger for establishing a network of persons and to deal with the transdisciplinary complex systems created [21] by the Fourth Industrial Revolution. The National Academy of Science and Engineering report [1], previously referred, is clear about the training needs of workforce in managing complexity, abstraction and problem-solving. It is expected that they are able to act much more on their own initiative and to possess excellent communication skills and the ability to organize their own work, putting greater demands on employees' subjective skills and potential. Previous reports such as the ones from American Society of Mechanical Engineers (ASME) and Royal Academy of Engineering [22, 23] alerted also to these needs. At the same time, new opportunities in terms of qualitative enrichment of their work, a more interesting working environment, greater autonomy and more opportunities for self-development will be provided.

3. Research methodology

To achieve the objectives of the study described in this chapter, two research instruments were used. Instrument 1 consisted in a questionnaire on the development of social skills, targeting IEM first-year students of the last PBL edition of the Industrial Engineering and Management (PBL-IEM1) program, at University of Minho, Portugal. This was aimed at uncovering the social skills that are developed and exercised in PBL, along with a reflection on its importance for future professional proficiency. Instrument 2 consisted in interviews on the perceived importance of early experiences in PBL for the development of social Skills, targeting a group of young Industrial Engineers currently working in a number of companies. These instruments enable the gathering of different perspectives from different stages of professional development, and the use of multiple research approaches and data types, that is, a quantitative analysis grounded on the interviews (via email) to IEM recently graduated engineers.

The questionnaire used on instrument 1 is part of a larger annual questionnaire on evaluation of the PBL methodology. Three sections were considered to directly relate to social skills development, which encompassed a total of 25 questions, which were evaluated based on

Section	Code	Question	
II. Learning and skills development	Q10	I feel that my participation in the PBL helped to develop my autonomy	
	Q11	During the semester, I improved my communication skills (oral and written)	
	Q12	Providing feedback to a report from another group was important and allowed to develop my critical thinking	
	Q13	The construction of the prototypes stimulated my capacity for initiative and creativity	
	Q14	Creating a blog/website was useful to help organize and disseminate the teams' project	
	Q15	The blog/website also helped to select and organize contents	
III. Teamwork	Q16	Teamwork has helped to increase my motivation for learning	
	Q17	I prefer to work in groups than individually	
	Q18	During the semester, I played an active role in the group	
	Q19	The existence of roles in the group (president, secretary, time manager) was fundamental to understand better those roles	
	Q20	During the project, my group held formal meetings and produced meeting minutes	
	Q21	I consider that the interpersonal skills that I have developed are important for my future professional activity	
	Q22	I shared the results of my tasks and knowledge with the rest of the group	
	Q23	I was able to solve the conflicts in the group and face them positively	
	Q24	I have applied teamwork supporting techniques that were useful to progress the project	
	Q25	I think that I played well as a team member	
	Q26	When in disagreement with the other team colleagues, we always came up to an understanding	
	Q27	After each meeting and task accomplished, I always left confident and enthusiastic	
	Q28	At each team meeting, I tried to understand the feelings of the others colleagues when they were angry, bored or sad	
	Q29	When I was not able to fulfill a task assigned to me, I asked for help to other colleagues of the group.	
	Q30	When a colleague completes the task correctly, I commend him for the achievement	
IV. PBL as teaching/learning methodology	Q48	The PBL facilitated my integration and socialization at the university	
	Q50	The PBL has a positive impact on the relationship established with teachers and the department	
	Q55	PBL requires excessive effort when compared to any other course of the semester	
	Q56	I believe, however, that this effort is rewarded with the skills acquired	

Table 1. PBL questionnaire: social skills development.

Position/Company type	IEM1 edition attended (Work experience)	Job responsibility	Role within teams Daily frequency of interaction
Trainee International	2012/2013 (finishing the fifth year)	Times and methods trainee engineer	Work in a multidisciplinary team. In average, work with six workers/ day.
Trainee International	2012/2013 (finishing the fifth year)	Times and methods trainee engineer	Work in a multidisciplinary team. In average, work with 20 workers/ day.
Trainee National	2012/2013 (finishing the fifth year)	Production	Lead a team of 30 workers In average, work with 10 workers/ day.
Trainee National	2012/2013 (finishing the fifth year)	Logistic and warehouse organization	Lead a team of nine workers In average, work with 40 workers/day.
Engineer/Trainee (abroad) International	2010/2011 (2 years)	Time Study and Methods engineer	Work in a multidisciplinary team. In average, work with 20–25 workers/day.
Engineer National (startup)	2010/2011 (2 years)	Industrial Engineering Marketing	Work in a multidisciplinary team. In average, work with five workers/ day
Engineer (abroad) International	2007/2008 (5 years)	Capability Acquisition Manufacturing Engineer	Lead one worker indirectly.
Engineer International	2006/2007 (6 years)	Industrial engineer	Work in a multidisciplinary team. In average, work with 50 workers/ day.
Engineer International	2006/ 2007 (6 years)	Process quality coordinator	Team moderator, three workers team multidisciplinary.
Engineer International	2004/2005 (8 years)	Logistic projects department responsible	Lead a multidisciplinary team of 8 direct and 16 indirect workers. In average, work with 50 workers/ day.

Table 2. Characterization of the interview respondents (young industrial engineers).

5-point Likert-type scale. The students had to indicate the degree of agreement to each question/sentence (where "1" corresponds to "strongly disagree" and "5" to "totally agree"). The questionnaire was made available online, for 2 weeks, after the end of the semester. From the 48 enrolled students, 32 accepted the challenge and answered the questionnaire. The 25 closed questions and the corresponding sections are depicted in **Table 1**.

Regarding the young engineers' opinion on the effects of the PBL methodology on the development of social skills, three main questions were raised and answered by e-mail:

- Did you feel that having carried out a project in the PBL context was relevant to your qualification (IME)? Explain in detail how?
- Do you think it helped you to develop social skills? Which ones? Can you please give examples?

• In the exercise of your professional activity did you feel that you were able to more easily overcome difficulties (or not) due to your knowledge acquired with PBL? Identify some of the difficulties encountered?

Six young engineers and four trainees voluntarily accepted to answer the questions. The sample was defined for convenience purposes, namely easiness and speedy gathering of the information. Concerning the work experience, four of them are having their first experience in an internship program (trainee) in a company, and the remaining ones from 2 to 8 years of work experience. Most respondents were integrated in multidisciplinary teams, with small and medium dimensions, and some of them were in a leading position (three). Two of the interviewees are, currently, working abroad in international company. **Table 2** characterizes the respondents' profile, company type, IEM first year of academic years (edition) attended and number of years of work experience, type of work and if they lead a team and how many workers they lead or work with.

A quantitative analysis of the results on the questionnaire and a qualitative analysis of the answers to the interviews will be explored in more detail using context analysis in Sections 5.1 and 5.2, respectively.

4. Study context

An interdisciplinary PBL approach was implemented in the first year of the Master's Degree on Industrial Engineering and Management (IEM) program, called PBL-IEM1. This approach was firstly employed in the 2004–2005 academic year [24]. The 14th edition of the PBL-IEM1 project was held in the first semester of 2016–2017 and had 49 students enrolled. This approach challenges the teams to develop and fetch knowledge, and skills, from different disciplinary fields, in order to propose a solution for a semester-wide open-ended problem. The class was divided into six teams of eight members each, with the exception of one team, which had nine members. The teams were initially instructed on the general guidelines of PBL and in detailed aspects of its particular implementation at the IEM program at the University of Minho. The dimension of the teams is kept large purposely, as well as for practical reasons, namely due to scarcity of project rooms, a limited number of available kits for prototype development and a limited number of tutors to accompany each one of the teams.

The first semester of the IEM program includes six Project Supporting Courses (PSCs), each of one holding five ECTS. The PSCs pertain to four departments, from two distinct schools, namely Engineering and Sciences, as shown in **Figure 1**. The PSCs are as follows: (1) Introduction to Industrial Engineering and Management (Topics of IEM); (2) Calculus; (3) General Chemistry; (4) Algorithms and Programming; (5) Linear Algebra and (6) Interdisciplinary Project on Industrial Engineering and Management. It involves a team of six lecturers and four tutors (two of the lecturers perform as a tutor as well), which are, simultaneously, the coordination team of the PBL edition.

The PBL-IEM1 lasts for 20 weeks (15 weeks of contact work), 2 weeks for Christmas break, and the remaining weeks are used for assessment purposes and final examination the PSCs. Although social skills can also be learned, regular practice and experience are keys for skills development. The PBL-IEM1 requires a great deal of contact, not only among team mates who



Figure 1. Distribution of IEM courses of first year, first semester by schools and departments [25].

share the same room (the teams' space) during a full semester but also a more frequent than usual interaction with lecturers and tutors. Frequent presentations to big audiences and communication with companies are also exercised during the semester. The teams are instructed as well on the theme that they have to develop [26], on the mechanism of peer assessment, on team working [27] and on conducting presentations.

The project assessment methodology (team related) is intricate, requiring the assessment of multiple items, each of them holding different weights and graded by multiple lecturers. More details on the PBL-IEM1 assessment methodology can be found in Moreira et al. [28–30], Fernandes et al. [31, 32] and Alves et al. [25, 33].

Previous research studies on PBL-IEM1 shown that the motivation to learn represents a prime driving force in Engineering Education which is influenced, not only on the interests and perceived importance of the subjects for the individual, but also on the "fun-factor" and on the inner dynamics of the individual teams, among other aspects [26]. The development of technical and transversal competences based on student-centered work is enabled through the PBL methodology and that the students themselves perceive that they have developed communication and teamwork competences [27], along with conflict management and commitment with others [34]. These findings suggest that active participation in real life should be stimulated during the learning process itself by enabling a meaningful student-centered learning environment. Moreira and Sousa [35] report an increased enthusiasm, cooperation and salutary competition between contending teams in the development of prototypes of production systems within PBL-IEM1. At the same time, teachers and researchers are also deeply involved with PBL process evaluation in a permanent cycle of continuous improvement, concerned with all aspects of PBL process [36].

5. Results and discussion

This section presents the results of application of two distinct approaches to evaluate the PBL effect on development of social skills. The results on the questionnaire, issued to first-year students on the 14th edition of the PBL-IEM1 (2016–2017 academic year), are provided in Section

5.1, while the results on the answers to the interviews, issued to recently graduated engineers, which undertook a prior edition of the PBL-IEM1, is provided in Section 5.2. The discussion on the foregoing matters follows the respective presentation of results on each one of the sections.

5.1. First-year IEM students' belief

Figures 2, **5** and **7** illustrate the distribution of the mean obtained based on first-year IEM students' perceptions regarding the social skills for Section II—Learning and skills development, Section III—Teamwork and Section IV—PBL as teaching/learning methodology, respectively. For all the statements in analysis, the mean obtained is greater than 3, showing a positive agreement. A detailed explication of the results of each figure is given next.

Although practically all the mean values obtained regarding Learning and skills development statement evaluation are around 4 (Agree) (**Figure 2**), the less positive agreement was obtained in Q15 (the blog/website also helped to select and organize contents) with 3.4.

Also, Q14 (creating a blog/website was useful to help organize and disseminate the teams' project) received a relatively low agreement (3.7). By these last two results obtained, it seems that the students do not hold a positive opinion on the use of a blog/website as a tool to improve their critique writing even in a less formal way [37]. **Figure 3** shows one of the blogs created by one team. Here it is possible to observe their concern to create an emotional appealing slogan related with their conceived company to employ an ecological material while denoting some creativity. Thus, they could not totally agree but they were enthusiastic in creating the blog and conveniently disseminate their work in progress. Nevertheless, in the fourth year, when they have PBL again, they recognize this value as they need it to demonstrate, and monitor, the results to teachers and companies [38].



Figure 2. Questionnaire results: Section II-Learning and skills development.



Figure 3. A blog created by one of the teams.

Concerning the Teamwork dimension (**Figure 4** shows two of the teams from 2016 to 2017 edition working in their project room), the graph of **Figure 5** shows that in average the students agree (around 4) with the statements considered, being Q27 (after each meeting and task accomplished, I always left confident and enthusiastic) the one that has received the lower value (3.4). In a way, students filling unconfident and not so keen on the work done indicate a certain inadaptability to adjust to changing situations, persons or/and environments. However, one such behavior could be expected, since it consisted a first-time experience for some of the students, if not for all. When asked about as they like to work, individually or as a group (Q17), students' responses in average show a slight agree. However, they also strongly agree (4.5) to share results of their tasks and knowledge with the rest of the group (Q22). Students in PBL context point out some weaknesses and strengths to teamwork, as discussed in Alves et al. [27].

The questionnaire of the Section IV—PBL as teaching/learning methodology also produced positive results (all of them equal or higher to four: "agree") (**Figure 6**).

The relationship established with the teachers (and other members of the department of production and systems) of the IEM program is so effective that starting in the first year the



Figure 4. Two teams working in their project room (pictures intentionally blurred for privacy purposes; taken by one of the authors).

Project-Based Learning and its Effects on Freshmen Social Skills in an Engineering Program 19 http://dx.doi.org/10.5772/intechopen.72054



Figure 5. Questionnaire results: Section III-Teamwork.



Figure 6. Questionnaire results: Section IV-PBL as teaching/learning methodology.

students are involved in the Department Day event, by presenting their projects (**Figure 7**) and by supporting and participating in the event.

Furthermore, some were even interested in starting their new experience in publishing papers [39] or in publishing their master dissertation final results in conferences and journals. A list of these publications can be seen in Alves et al. [40]. At the same time, their activity in alumni associations shows a dynamism toward to increase their collaboration on the international



Figure 7. First-year students' teamwork presentations in the Department Day (pictures intentionally blurred for privacy purposes; taken by one of the authors).

association of Industrial Engineers students (ESTIEM). Their activity is so intense that they were invited to become the magazine editors, while the number of participants in the international events was never so high.

Also, the mean value near 5 ("Strongly agree") obtained in statement Q55 (PBL requires excessive effort when compared to any other course of the semester) strengthens the difficulty that the students experience in managing time. But, at the same time, they also gave a good grade to the Q56 (I believe, however, that this effort is rewarded with the skills acquired). That is, they noticed that PBL is not equal to other curricular units.

From the 32 responses obtained in the questionnaire, it was also possible to retain some phrases reported by the students in the open question "Positive aspects of participating in the PBL":

- Development of soft skills and better understanding of the complexity of a company/industry, namely, at the level of its production system.
- Development of the spirit of initiative and "gain" more responsibility.
- Socialization with colleagues; learning of transversal competences; similarities to an employment environment (working with large groups and knowing how to manage the work and opinions of each one).
- Teamwork and resolution of various problems in a short time; acquisition of new skills/soft skills; teamwork; acquisition of knowledge that goes far beyond what is given in each course.
- The PBL allowed to develop the soft skills and the ability to work as a team; to adopt a better perspective on the course and future professional reality; to improve our ability to identify and solve problems; and to deal with different opinions.
- Skills acquired in presentations, in writing a report, in group work, in being more autonomous and able to distribute tasks and adapt to conditions.
- To learn to deal with different personalities, and sometimes they clashed with ours; in this sense, in order to keep the group going, we had to grow as people and try to understand the others as much as possible.

- There was greater proximity between teachers and students, which contributed to a better overall environment.
- Personal enrichment, coupled with the development of highly diversified skills.
- The various and different presentations throughout the semester helped us to improve our oral communication and, above all, helped us to face the fear of speaking to many people. Aspects such as creativity and initiative were greatly stimulated, as well as the development of the critical spirit.
- This project was undoubtedly enriching at all levels; however, it should be noted that the initial shock was great, everything at first seemed impossible to materialize! During the semester, the work was arduous and exhausting, but of course there is no finest rewarding sensation than what we felt; it is something that is difficult to explain in words...

5.2. Young engineers' opinions

Regarding the young engineers' written answers to the set of posed questions, the analysis will be presented below.

Attending to the first question: "Did you feel that having carried out a project in the PBL context was relevant to your qualification (IEM)? Explain in detail how?"

The qualitative analysis using the webQDA software [41] produced the result shown in **Figure 8**. It can be seen that projects are important as *"the key to the development of future indus-trial engineers"* as mentioned by one young engineer with 5 years of work experience in an automobile industry company that holds a recognized and strong brand.

The second question was: "Do you think it helped you to develop social skills? Which? Can you please give examples?" **Figure 9** depicts the results.

To develop and sharing ideas, to acquire different skills, to deal with others feelings, to do the presentations and to develop communication skills, among others, are competencies difficult to achieve in lecture-based classes [15].



Figure 8. Young engineers' most frequent words for the question 1.



Figure 9. Young engineers' most frequent words for the question 2.

Figure 10 presents the results of the third question: "In the performance of your professional activity did you feel that you were able to overcome more easily the difficulties (or not) due to your knowledge acquired with PBL? Identify some of the difficulties encountered?"

In this case, one of the words most identified was "problem," however, with a positive mean. As mentioned by one of the young engineers:

"Obtaining relevant information for analysis and problem solving: working in different teams with different personalities and solving problems."



Figure 10. Young engineers' most frequent words for the question 3.

6. Conclusions

This chapter presents a study about the effect of the PBL on social skills development of first-year Industrial Engineering students. The instruments used provided evidence that PBL programs deliver a rich context where social skills can effectively be exercised and enhanced. A number of such skills were identified, and the development process thoroughly discussed, namely, effective team working, conflict management, effective oral and written communication, capacity of adaptation to different work environments, assuming responsibilities, caring about the other's learning, assessing the work of others and of their own, willingness to pertain and work in activities of student and professional associations. The acquired competences allow students to grow and become professionals capable of leading and working in projects and multidisciplinary teams.

Acknowledgements

This work has been supported by COMPETE: POCI-01-0145-FEDER-007043 and FCT – Fundação para a Ciência e Tecnologia within the Project Scope: UID/CEC/00319/2013.

Author details

Anabela C. Alves*, Celina P. Leão, Francisco Moreira and Senhorinha Teixeira

*Address all correspondence to: anabela@dps.uminho.pt

Department of Production and Systems, Centro ALGORITMI, University of Minho, Guimarães, Portugal

References

- [1] Kagermann H, Wahister W, Helbig J. Recommendations for implementing the strategic initiative INDUSTRIE 4.0: Securing the Future of German Manufacturing Industry. Final report of the Industrie 4.0 Working Group, April 2013. p. 112
- [2] Kahlen F-J, Flumerfelt S, Alves CA. Transdisciplinary Perspectives on Complex Systems: New Findings and Approaches. Switzerland: Springer International Publishing; 2017
- [3] American Management Association (AMA), Critical Skills survey. 2012. Available: http// www.amanet.org/uploaded/2012-Critical-Skills-Survey.pdf
- [4] UNESCO. Engineering: Issues, Challenges and Opportunities for Development. UNESCO Report; UNESCO Publishing; 2010

- [5] BusinessDictionary. Social Skills. 2017. [Online]. Available from: http://www.businessdictionary.com/definition/social-skills.html [Accessed: 26 Mar 2017]
- [6] Collins. Social Skills. 2017. [Online]. Available from: https://www.collinsdictionary.com/ us/dictionary/english/social-skills [Accessed: 26 Mar 2017]
- [7] Bonwell CC, Eison JA. Active Learning: Creating Excitement in the Classroom. Washington, DC: Wiley; 1991
- [8] Mills JE, Treagust DF. Engineering Education Is Problem-based or Project-based Learning the Answer? Australasian J of Eng Educ, online publication 2003-04. p. 16. Available: http://www.aaee.com.au/journal/2003/mills_treagust03.pdf
- [9] Graaff E, Kolmos A. Management of Change: Implementation of Problem-Based and Project-Based Learning in Engineering. Rotterdam: Sense Publishers; 2007
- [10] Schmier L. Random thought: A sacred trust. Atwood Publishing. 2001. [Online]. Available from: http://www.halcyon.com/arborhts/rt/01may23.htm. [Accessed: 23 Apr 2017]
- [11] Gibbs G, Habeshaw T. Preparing to Teach: An Introduction to Effective Teaching in Higher Education. 2nd ed. Bristol: Technical & Educational Services Ltd; 1992
- [12] Krathwohl DR. A revision of Bloom's taxonomy: An overview. Theory Into Practice. 2002;41(4):212-218
- [13] Fink LD. Creating Significant Learning Experiences: An Integrated Approach to Designing College Courses. 1st ed. New York: Jossey-Bass; 2003
- [14] Flumerfelt S, Kahlen F-J, Alves AC, Siriban-Manalang AB. Lean Engineering Education: Driving Content and Competency Mastery. New York: ASME Press; 2015
- [15] Prince M. Does active learning work? A review of the research. Journal of Engineering Education. 2004;93(3):223-231
- [16] Knoll M. The project method: Its vocational education origin and international development. Journal of Industrial Teacher Education. 1997;34(3):59-80
- [17] Dewey J. Democracy and Education. An Introduction to the Philosophy of Education. New York: Free Press; 1996
- [18] Kilpatrick WH. The project method. Teachers College Record. 1918;19(4):319-335
- [19] Kilpatrick WH. Dangers and difficulties of the project method and how to overcome them. Teachers College Record. 1921;22(4):283-287
- [20] Powell P, Weenk W. Project-Led Engineering Education. Utrecht: Lemma Publishers; 2003
- [21] Kahlen F-J, Flumerfelt S, Alves A. Transdisciplinary Perspectives on Complex Systems. Switzerland: Springer International Publishing; 2017
- [22] ASME Board on Education. Creating the Future of Mechanical Engineering Education: An Action Agenda for Educators, Industry, and Government ASME Board on Education. Report, September 2012. p. 4
- [23] Graham R. Achieving excellence in engineering education: The ingredients of successful change. Report, March 2012. p. 70
- [24] Lima RM, Carvalho D, Flores A, Van Hattum-Janssen N. A case study on project led education in engineering: Students' and teachers' perceptions. European Journal of Engineering Education. 2007;32(3):337-347
- [25] Alves AC, Sousa RM, Fernandes S, Cardoso E, Carvalho MA, Figueiredo J, Pereira RMS. Teacher's experiences in PBL: Implications for practice. European Journal of Engineering Education. 2016;41(2):123-141
- [26] Moreira F, Mesquita D, Van Hattum-Janssen N. The importance of the project theme in project-based learning: A study of student and teacher perceptions. In: Proceedings of the 2011 Project Approaches in Engineering Education. Vol. 53(9) (digital edition). 2011. pp. 65-71
- [27] Alves AC, Mesquita D, Moreira F, Fernandes S. Teamwork in project-based learning: Engineering students' perceptions of strengths and weaknesses. In: International Symposium on Project Approaches in Engineering Education (PAEE2012), (digital edition). 2012. pp. 23-32
- [28] Moreira F, Fernandes S, Malheiro M, Ferreira C, Costa N, Rodrigues C. Assessing student individual performance within PBL teams: findings from the implementation of a new mechanism. In: "Global Research Community: Collaboration and Developments", Proceedings of the 5th International Research Symposium on Problem Based Learning (digital edition). 2015. pp. 35-47
- [29] Moreira F, Sousa R, Leão, CP, Alves AC, Lima RM. Project-Led engineering education: Assessment model and Rounding Errors Analysis. In: Proceedings of 3rd International Conference on Integrity, Reliability and Failure. Porto: Portugal; (digital edition). 2009. p. S0233_P0552
- [30] Moreira F, Sousa RM, Leão CP, Alves AC, Lima RM. Measurement rounding errors in an assessment model of project led engineering education. Journal of Online Engineering. Nov 2009;5(S2):39-44
- [31] Fernandes S, Flores MA, Lima RM. Students' views of assessment in project-led engineering education: Findings from a case study in Portugal. Assessment & Evaluation in Higher Education. 2012;37(2):163-178
- [32] Fernandes S, Flores MA, Lima RM. Student assessment in project based learning. In: Van Hattum-Jansen N, Campos LC, Dirani EA, Manrique A, editors. Project Approaches to Learning in Engineering Education: The Practice of Teamwork, Chapter 9; Rotterdam: Sense Publishers; 2012. pp. 147-160
- [33] Alves A, Sousa R, Moreira F, Alice Carvalho M, Cardoso E, Pimenta P, Malheiro MT, Brito I, Fernandes S, Mesquita D. Managing PBL difficulties in an industrial engineering and management program. Journal of Industrial Engineering and Management. 2016;9(3):586-611
- [34] Alves AC, Moreira F, Lima R, Sousa R, Dinis-Carvalho J, Mesquita D, Fernandes S, Van Hattum-Janssen N. Project based learning in first year, first semester of industrial

engineering and management: Some results. Education and Globalization; General Topics. 2012;5:111

- [35] Moreira F, Sousa RM. Development of production system prototypes in the context of interdisciplinary Project Based Learning (in portuguese). Desenvolvimento de protótipos de sistemas de produção no âmbito da aprendizagem baseada em projectos interdisciplinares. In: Proceedings of the 5th Congresso Luso-Moçambicano de Engenharia. Porto: Edições ENEGI; 2008. pp. 03A004.1-8
- [36] Alves AC, Leão CP. Action, practice and research in project-based learning in an industrial engineering and management program. In: Proceedings of the ASME 2015 International Mechanical Engineering Congress & Exposition (IMECE2015). ASME Digital Collection; 2015. pp. V005T05A013
- [37] Marsden N, Piggot-Irvine E. Using blogging and laptop computers to improve writing skills on a vocational training course. Australasian Journal of Educational Technology. 2012;28(1):30-47
- [38] Vicente S, Mattarredona E, Alves AC. The importance of blog as a communication tool to support the development of project-based learning. In: Proceedings of International Symposium of Project Approaches (PAEE2014), (digital edition). 2014. pp. ID39.1-9
- [39] Ramires F, Martins M, Cunha M, Alves AC. Different structures of projects in engineering: The perspective of freshmen students. In: 8th International Symposium on Project Approaches in Engineering Education and Active Learning (PAEE'ALE2016) (digital edition). 2016. pp. 661-669
- [40] Alves AC, Sousa RM, Dinis-Carvalho J, Moreira F. Lean education at University of Minho – Aligning and pulling the right requirements geared on competitive industries. In: Alves AC, Flumerfelt S, Kahlen F-J, editors. Lean Education: An Overview of Current Issues. Switzerland: Springer International Publishing; 2017. pp. 149-176
- [41] Souza FN, Costa AP, Moreira A. O Manual de Utilizador de um Software de Análise Qualitativa: as perceções dos utilizadores do webQDA. RISTI - Rev Ibérica Sist e Tecnol Informação 2016:107-117

Competences in Project Management: A Case Study in Osaka Institute of Technology

Makoto Katoh, Yutaka Kawata, Toshio Haga, Hiroyuki Kobayashi, Tsutomu Yoshimura, Kazuo Kumamoto, Muneyoshi Iyota and Keiko Natori

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.72338

Abstract

This chapter briefly explains the distributed project management of four joint departments for first-grade joint project-based learning (PBL). That is, conventional studies and purpose, capital, competence and ability of four kinds of distributed project management were presented, and relations among them are shown after describing background (included prehistory of PBL in Osaka Institute of Technology (O.I.T.)). Then, consideration and analysis about communication, interaction, cooperation, merit, week-point, effect and inference were discussed. Some case studies were described about open innovation and competences in stakeholder management. They were distinctive and superior in first-grade PBL of O.I.T. Finally, some future themes were presented.

Keywords: project management, competence, case study, open innovation, stakeholder

1. Introduction

First, project-based learnings (PBLs) in Japan, prehistory and outline of PBL in Osaka Institute of Technology (O.I.T.), conventional studies and the contents of each section are described in this section. A project is a job having a new element and a certain period (start and end), a work evolving gradually with a clear purpose to be done, influencing the object, and manipulating the object. Competence has various definitions depending on the field but based on the world standard of competence of project manager here, and it makes unique meaning for each distributed project management. Then, it must be used well in the society.



© 2018 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. Then, project-based learning (PBL) has been spreading into the world of education because the project-based learning method can develop and improve the competence elements for finding problems, understanding problems and solving problems of the students because the overall thought process for learners in active learning by team-based learning (TBL) is important.

For this, the teachers of four departments including one leader in each class room (The parts of the leaders are project managers who show later in Section 2) should make the structure welldefined about major large problems and student missions by using various competences in the project management.

Here, the project management makes the initial purpose to clear and progress so that it gives skill, technique and tools for achieving the requirements (quality, cost, demand and range of products) of the project. It makes the specifications, plans and methods to clear for requirements by many stakeholders. Moreover, project management has a SPDCAF cycle, that is, Start, Plan, Do, Check, Action, Finish. Especially, start phase attached to a known PDCA cycle is important because it moves the project from idea phase to plan phase.

The literatures about PBL in Japan were seen in Table 1.

The PBL type learning was introduced to Japan as "Engineering Design" by Accreditation Board for Engineering and Technology (ABET). Japan Accreditation Board for Engineering Education (JABEE) advised in 2004 that the Japanese engineering education had only analytic subject and no subject for synthesizing.

Then, JABEE introduced "Engineering Design" as an essential subject to accept by JABEE accreditation. Japanese university with faculty of engineering has introduced "Engineering Design" since that time. This must be further paid attention as a very effective way for active learning. Here, active learning is a study technique by which the students are able to find problems independently, considering the solution for the problems and solving the problems.

In Mechanical Engineering Department of Osaka Institute of Technology (O.I.T.), JABEE had been examined from 2000, and PBL was introduced as a new subject in the third grade (shown in **Table 2**) in 2008 for recognition of JABEE.

After then, the official PBL was spread to second grade and first grade, although the informal PBL was tried in fourth grade of partial laboratory as shown in **Table 2**.

No.	Affiliation	Contents					
[1]	Nitobe School	Results of teaching method for the first-grade since 2015 are presented, and findings through this work were shared. In Nitobe School curriculum, there are another education courses (Start-up, Global Issues, Problem-solving and Problem-finding) as four terms (spring, summer, fall and winter).					
[2]	Hokkaido University of Japan	Comparison of student attitude by PBL for radiological technologists was introduced.					
[3] Keio University Media Institution		Education for software engineers through university and industry collaboration was done.					

Table 1. Example about PBL in Japan.

Course	Term	Prehistory and outline					
First-grade PBL	First grade	This started as four departments joint PBL. The Departments of Mechanical Engineering, Electric Engineering, Electronics Engineering and Robotics Engineering jointly conduct the half-year PBL for freshmen. The theme is robot car with multifunction. It started in 2014.					
International PBL	Second grade	This is to establish first international PBL between National Taipei University of Technology and O.I.T. It was carried out since 2013 between Electric, Electronics and Mechanical Departments of O.I.T. and Mechanical Department of NTUT. The program is consisted of 1 week of intensive cooperation with five mixed team of students. The theme was robot car and wind turbine with wind lens.					
Engineering Practice	Third grade	Mechanical Engineering Department introduced Engineering Practice as an Engineering Design subject. This subject is designed as the 1 year PBL of the design and manufacturing consist of six topics: electric vehicle, hovercraft, engine and motor cycle, robot, steel can recycle machine, and so on.					
Graduation Research	Fourth grade	In partial laboratories of Mechanical Engineering Department, few graduate researches had been done like PBL and TBL before/after starting of Engineering Practice as third-year PBL.					

Table 2. Outline of PBL in O.I.T.

The outline of PBL in O.I.T. is summarized in Table 2.

Nowadays, a project and program management (P2M) model in PBL has become important. Here, the program is a set of several projects assembled and integrated. Then, all the abovementioned PBL in four grades had better to be managed as a program.

No.	Authors	Contents			
[4]	Baumgardner CR, Shane GS, Grant KP	Leadership competences of sustainable construction project management have been explained in literature. The study has generated a new model to facilitate the process of sustainability in the industry and extends some of the significant components from leadership assessment in the context of construction project management in sustainable building projects.			
[5]	Amin Akhavan Tabassi, Kamand M. Roufechaei, Mahyuddin Ramli, Abu Hassan Abu Bakar, Radzi Ismail, A. Hamid Kadir Pakir	It has been considered Architecture Management, Economics Evaluation of Project, Profile of Production & Risk Control, Resource Procurement, Competence of Project Management in Project and Program Management (P2M) in Japanese drug development			
[6]	Iwasaki K	Technical competence was perceived to be more important for the management of extremely good teams than it was for the management of reasonably good teams.			
[7]	Saito T	"Men, money, and material" is usually pointed out in Japan, as capitals of a general project management.			
		The literature enlarges to five capitals, added "Information and technique" to these. In addition, they showed five competences of "Foreign adjustment competence, risk control competence, internal control competence, construction management competence, and technical improvement competence."			

Table 3. Conventional studies about project management.

This section mainly focused on this joint four department PBL and the distributed project management for the Mechanical Engineering Department out of joint four departments. Here, distributed project management is a new management technique by distributed project managers who consigned partially by the main project manager.

Next, Table 3 presented conventional studies about project management.

The purpose of this chapter is to present about competences of individual distributed project management as in the case of Mechanical Engineering Department. Moreover, it is to show image of an ideal project management as one and a certain kind of ideal distributed project management taking account of the results.

Section 2 explains competences in distributed project management about relation, competences in each (main, sub, boss, next) project management and evaluation.

Section 3 presents consideration and analysis about "communication, interactions and cooperation," "merit, week point and effect, inference," "an Ideal project management as one" and "ideal distributed project management."

Section 4 explains case study of open innovation by teamwork of enlargement meeting obtained as results of the distributed project management.

Moreover, Section 5 shows competences in stakeholder management including the open innovation, weekly report and joint competition. Finally, Section 6 presents the conclusion.

Especially, Refs. [5, 6] are a drug field, P2M is being handled, and its contents are interesting.

2. Competences in distributed project management

This section presents relation, competences and evaluation about distributed project management.

2.1. Distributed project management

2.1.1. Structure of relation on major project management

First of all, aim and purpose of distributed capital project management group (*Boss, Main, Sub, Next, Stakeholder*) are defined for future integration and redistribution. Although not all project structures always have the same stakeholders (*Boss, Main, Sub, Next*), stakeholder analysis and leadership described in this section are the most usual relationships. We must justify the relation to stakeholders for open innovation.

Aim: Four *Main* are defined for major management of the project by major project managers of four departments. First, the joint meeting was formed by them. *Sub* is defined for support and assist of major project management by sub project manager. *Next* is defined for project management by next generation project managers in the project. *Boss* is defined for project management by boss of the project who proposed the project.

Competences in Project Management: A Case Study in Osaka Institute of Technology 31 http://dx.doi.org/10.5772/intechopen.72338



Figure 1. Relation among repeated distributed project management.

Though generally speaking, *boss* of the project or *BOSS* is disappeared from the project after proposing the project, he or it has been appeared again as a general teacher for evaluation and feedback function of the project in this case as shown in **Figure 1**.

Personnel purpose: *Mains* work to lead the project to success cautiously while being afraid of failure, by using power of members in some cases even if it held himself/herself down. *Sub* works to support and assist the main project management and innovates the joint project without fear of failure. *Next* works to learn how the project proposed will be successful. *Boss* works to relearn why the project proposed by him/her was successful.

Figure 1 shows relation among the four kinds of repeated distributed project management (*Boss, Main, Sub, Next, stakeholders i*) by managers (*boss, main, sub, next, stakeholders i*).

Competences (ability to do well) in each distributed project management and relation among repeated distributed project management are explained in the following paragraphs.

2.1.2. Stakeholders management

2.1.2.1. Analysis

For the students in each year, as most important stakeholders changed as every project in this PBL, the individuals of each group decide their roles and group leader through hope by early investigation in the project; then, they have identified their interests by the investigation [8]. Moreover, their interests and influence to the group with students of the other departments can be identified by the teacher's comments in their weekly reports. Original items for the stakeholder analysis are as follows:

- Target of this week (group and individuals).
- Content of execution of the entire group of this week.
- Content of execution of the individuals (inside class time and outside class time).
- Content of discussion in combination meetings.
- Achievement level, reflection point and improvement idea.
- Problem and work schedule for next week.
- Process of the future.

2.1.2.2. Leadership

The definition, purpose and description, and so on of leadership are presented in [8].

Here, the definition of leadership is providing direction and guidance to students and groups. It involves the ability to choose and apply appropriate styles of project management in different situations. Besides displaying leadership with his or her team, the student needs to be seen as a leader in representing the PBL to senior management and other interested parties [8]. Original items for leadership education are indicated in Section 2.6.

2.2. Competences in main project management

Competences in project management for main project managers are at least to seize three pillars of capital "Person, thing and money" and "Personnel right" and "Budget authority", involving the process of executing the responsibility of work allotment to each department. Moreover, they may have competences to present topics in a conference, decide on the schedule, settle the discussion and draw conclusions as chairpersons or sub-chairpersons.

2.3. Competences in subproject management

The project management competences of one subproject manager are based on experience knowledge and through dialogue with other project managers. He understands the purpose and contents of the whole project, proposing the technology necessary for the project competitively with them. He is also preparing explanatory materials for the students until the project gets on track.

2.4. Competences in boss-project management

Competences in project management for boss project manager who conceive the project are personnel ability that can choose talented person who was suitable for content of each project as project manager and are student and teacher's commending abilities, practical work with subproject manager, rich human nature and time management ability.

2.5. Competences in next generation project management

The competences of project management by young next generation project manager can organize the information of the meeting as minutes and can organize the outcome information of the project cleanly without changing its contents and make an external announcement. In consideration to detail, there are also external viewpoints. They were presenting methods of PBL at external meetings [9, 10].

2.6. Competences in stakeholders management

Social basic competences expected in stakeholders of the PBL are taught like the simple followings:

1. Problem solution

The work that fills necessary functions can be made.

2. Team work

They should know members and their ability, and they can help each other.

3. Communication

It puts in two cents hearing others' opinions.

4. Leadership

All members become leaders of something, and their roles are decided for that as the leaders.

5. Creativity

The aspect is changed, and they look at multipronged.

6. Time management

The progress report and the delay measures of the process deciding are managed.

2.7. Evaluation of distributed project management

1. Load reduction of main project manager.

The load of the MAIN increases oppositely, while the load of the SUB has decreased. This might be a result of special circumstances in case of O.I.T.

2. Sufficient number of leaders of classroom.

The member of expansion joint meeting was able to be arranged as at least one leader in each classroom.

3. External sending by various viewpoints.

Three external sending by each different viewpoint have already been done.

4. Effect of feedback by BOSS.

This is not well-understood still.

The next section describes consideration and analysis for the above managements.

3. Consideration and analysis

Moreover, this section adds the following consideration and analysis for the distributed project managements.

3.1. Communication, interactions and cooperation among distributed project management

Interaction, communication, conversation and discussion are not same terms but similar terms [11]. Then, cooperation are noticed as one obtained by human-computer interaction [12]. Here, it is noticed that interactions and communication can draw cooperation.

After observing the proposed method of distributed project manager compared with an ordinary project manager, communication, interaction and cooperation, which are selected as most important matters, were shown in **Table 4**.

3.2. Merit, week point and effect, inference analysis of distributed or decentralized project management method

Effect, inference and yield in failure mode or action mode, and so on were used for analysis of various systems in conventional study [13].

Here, **Table 5** shows merit, week point, effect and inference of proposed method of distributed project management compared with an ordinary project management briefly, although they were old and ordinary methods.

3.3. An ideal project management as one

The following item and six contents, which were integrated as ideal management of one and will cause a lot of main management of project, can be obtained after seeing the project management in which capital, rights and competences were distributed.

Table 6 shows a certain kind of ideal project management as one.

This will be important as a consensus problem in multiagent optimization systems in the future, although these contents do not obtain the consensus [14].

3.4. An ideal distributed project managements

Achieving an ideal abovementioned project management alone is difficult, the example of trying ideal four distributed project management is shown in **Table 7**.

This may be approaching to an optimal distributed project management, although this is not different from real distributed management.

Item	Contents
Interactions	There are both positive and negative interactions among the distributed project managers.
Communication	The chance of conversation between PMR increases, although there is also miss-over order [13] from <i>Main</i> or <i>Sub</i> to others in their communications.
Cooperation	They can draw cooperation by sharing the same target consideration.

Table 4. Interactions and communication can draw cooperation.

Item	Contents
Merit	There is a possibility that they can do something, which one project management is unable to do. They can reduce the load of main project managers.
Week point	Human cost up. However, it may not be a problem in the special environment like organizations for education.
Effect	Next grows up and the evaluation rises.
	All distributed project managers are able to know the project.
Inference	Spread of upper-grade PBL or another PBL after long time.

Table 5. Merit, week point, effect and inference.

Item	Contents
Five capitals	"Persons, objects, time, space and money"
Five rights	"Personnel right, budget authority, scheduling right, space securing right and greeting"
Five abilities	"Innovation of technology, control of information, information sending to inside and outside after taking and settling, agency to various distributed managers, to honor"
Five natures	"Cooperative, competitive, charismatic, responsibility and sympathy"
Five senses	"Selecting themes, material, parts, terms and methods"

Table 6. A certain kind of ideal project management as one.

Item	m Boss Main		Sub	Next/part		
Capitals		"Persons, objects and	"Time, space"			
		money."	"Information, technology"			
Rights		"Personnel right, budget authority and greeting"	"Scheduling right and space securing right"			
Abilities	"To honor and information sending to outside"	"Control of information, information sending to inside and *"	"Innovation of technology, control of information, information sending to inside and outside after taking and settling, agency to various distributed managers"	"Control of information, information sending to inside and outside after taking and settling"		
Natures	"Cooperative, charismatic, sympathy"	"Cooperative, competitive, sense of responsibility, sympathy"	"Cooperative, competitive, sympathy"	"Cooperative, sympathy"		

Table 7. A certain kind of ideal four distributed project management.

It seems that repeating while deliberating integration and decentralization like this is useful for the optimization of the system that does not use the expression.

This may be necessary in the future theme as a distributed optimization problem [14].

In the next section, we will explain the open innovation by team works of project managers, including openly distributed departmental and interdepartmental project management.

4. Open innovation by teamwork

This section presents open innovation by teamwork of project managers and stakeholders.

Open innovation is done in teamwork, while closed innovation is done individually.

4.1. Case studies

In multicultural engineering [15], a few open innovations [16] were tried by teamwork in project management. Some case studies are described in the following paragraphs.

4.1.1. Case 1

4.1.1.1. Procedures

- 1. Awareness of scenario valued for Mechanical Engineering Department team in joint four department of PBL from a symbolic object by subproject management. (Affordance of the object: Slope course like hill excess as shown in **Figure 2**).
- 2. Proposal of innovation by subproject management.
- **3.** Support of innovation by main project management and stakeholder management (especially for TA).

Examples of products: a mechanical sensor, controller and actuator on a car for throwing a ping pong gem from the box out of the goal wall without electric power source (**Figure 3**).



Figure 2. Courses with slope like hill where model cars run.

Competences in Project Management: A Case Study in Osaka Institute of Technology 37 http://dx.doi.org/10.5772/intechopen.72338



Figure 3. Configuration of mechanical position ON-OFF control with position detecting SW.

4.1.2. Case 2

- 4.1.2.1. Procedures
- 1. Motivation of innovation considered load balance by enlarged project management
- 2. Challenge and change of innovation by subproject management
- **3.** Awareness of culture and value of innovation by main project management (message between the project managers in joint meetings)
- **4.** Support of innovation by main project management and stakeholder management (especially for specialists and experts)

Examples of products: a set of noncontact feeding coils to transmit and receive the high-frequency power for detecting the goal wall (**Figure 4**).

In the next section, competences in stakeholder management are presented including open innovation.



Figure 4. Noncontact feeding twin coil.

5. Competences in stakeholder management

This section presents competences in stakeholder management for open innovation, weekly report and joint competition.

5.1. Competences in stakeholder management for open innovation

In Case 1, TAs instructed by a main project management were active as stakeholders.

In case 2, teachers asked by a main project management who did not participated in the project as stakeholders who supported the project management. To summarize, it was recognized that not only teamwork but also stakeholder management including the section manager plays an important role in the success or failure of the project.

5.2. Competences in stakeholder management for weekly report

The weekly report was imposed on the administrators to properly modify the form in each department. By letting students write a report in this form, they can grasp the common problems of all divisions and individual problems of each division and draw out the direction of student behavior for problem solving. Competences for making format of weekly report may be knowledge to teach the PBL and experience for incompleteness of the first-grade students' description matter for past weekly reports.

5.3. Competences in stakeholder management for joint competition

There are two cases of innovative technology by a main project management. In both cases, technical competencies for providing web address to upload word file or excel file, information management for settling important rules and scores of competitions are needed.

5.3.1. Case 1: explanation function of joint competition in orientation

The following specification of action flow chart expression (**Figure 5**) was most innovative and easy to understand for stakeholders including mechanical department.

Grey boxes mean the charge of mechanical department. Competences for making a chart like this chart may be knowledge on sequential flow chart using sequence control.

5.3.2. Case 2: online monitoring of joint mission and recent rules for joint competition

Because the rules for competition were changed by circumstances, they are managed by main project management so that wireless LAN of PC of each classroom may monitor the mission and the rules online though the joint mission and the rule for joint competition were explained in the orientation at first. Competences for making a system like this system may be knowledge about information and communication in PC. An example is shown in **Figure 6**.

5.3.3. Case 3: online monitoring of score book in joint competition

Because the scores of each team of other class rooms in competition were not known, they were managed by main project management so that teachers can change and monitor the

Competences in Project Management: A Case Study in Osaka Institute of Technology 39 http://dx.doi.org/10.5772/intechopen.72338



Figure 5. Action flow chart as explanation function of joint competition.



Figure 6. An example of online monitoring of joint mission and recent rules for joint competition.

scores, and other stakeholders will be able to monitor them through wireless LAN of PC in each classroom. Competences for making a system like this system may also be the same as the above system. Specification of online score book is shown in **Table 8** for reference.

Class room	Group and try no		Running time (s)	Work enter	Ping pong gem enter	Penalty at wall		Bonus at goal		Total score	
			to top	to goal	in the box		Collision to wall	Stop car	Light LED	Stop wheel	
151	1 Group	1									
		2									
	2 Group	1									
		2									

Table 8. Specification of online score book.

6. Conclusion

The distributed project management method for the first-year PBL of joint departments was proposed and considered about some important items after presented each distributed management and shown relations among the distributed managements.

The motivation leading up to this topic is the discovery of the book [14]. The problem of integrating distributed management would be a consensus problem, and there was a speculation that the method of distributed optimization could be used when decentralizing the integrated management.

A large advantage of such a distributed and decentralized project management method mutually supplements various project managements, goes up the perfection, executes a role each other and decreases the load concerning the project if it is compared with a conventional project management method.

It is important that such a new attempt concerning the project management of the Mechanical Engineering Department is the one approved by the management of another department. (It is because those advantages are shared.) Such an attempt does not succeed without fail, and it greatly depends on the competences of the member who participates in the same meeting. Therefore, it does not necessarily succeed even if the same thing is done by another project.

Some case studies were described about open innovation. It was motivated by which the hill excess course of model car makes to associate the blackout measures for accident of nuclear power generation caused by tsunami.

Then, competences in stakeholder management were discussed. They were distinctive and superior in first-grade PBL of O.I.T.

There are some future themes when you examine such context closely as follows:

- **1.** It is necessary to clarify the competences by thinking about the ability of each distributed project management for clarification of conditions for the application of such distributed project management to other projects.
- **2.** Problems derived from optimal controllers of projects ranging from ideal optimized project management to distributed project management [14].
- **3.** Optimal consensus problem for an ideal optimized project management by distributed project management [14].
- **4.** More appropriate lecture about project management may be necessary for first-grade students [17, 18].

Acknowledgements

The authors sincerely express their gratitude to Special Duty Prof. Keiichi Hirako of Keio University who lectured about the project of artificial satellite. They are also grateful to Prof. Koike, Prof. Miyabe, Part Lecturer Matsumoto and Associate Professor Yoshida who served as the teacher and member of meetings of Mechanical Engineering Department in the PBL for first-grade students and many stakeholders who manage or cooperate the PBL as deans, successive section chiefs, other project managers (Prof. Izuru Nishikawa), staffs, TAs and students.

Author details

Makoto Katoh*, Yutaka Kawata, Toshio Haga, Hiroyuki Kobayashi, Tsutomu Yoshimura, Kazuo Kumamoto, Muneyoshi Iyota and Keiko Natori

*Address all correspondence to: makoto.kato@oit.ac.jp

Osaka Institute of Technology, Osaka, Japan

References

- Ravankar AA, Imai S, Shimamura M, Chiba G, Takasuka T. Problem-based learning and problem finding among university graduate students. Journal of Higher Education and Lifelong Learning. 2017;24:9-20
- [2] Terashita T, Tamura N, Kisa K, Kawabata H, Ogasawara K. Problem-based learning for radiological technologists: A comparison of student attitudes toward plain radiography. BMC Medical Education. 2016;16(1):16-236
- [3] Matsuzawa Y, Oiwa H. A result of trial education for software engineers through university-industry collaboration and project-based learning. Transactions on Signal and Information Processing Association. 2007;48(8):2767-2780
- [4] Baumgardner CR, Shane GS, Grant KP. The perceived importance of technical competence to project managers in the defense acquisition. IEEE Transactions on Engineering Management. 1997;44(1):12-19
- [5] Tabassi AA, Roufechaei KM, Ramli M, Bakar AHA, Ismail R, Pakir AK. Leadership competences of sustainable construction project managers. Journal of Cleaner Production. 2016;124:339-349
- [6] Iwasaki K. Research of business model as high-risk & high return on new drug development. Journal of International Association of Project & Program Management. 2008;2(2):79-88 (in Japanese). DOI: http://doi.org/10.20702/iappmjourante.2.2_79
- [7] Saito T. Primary study on nature and abilities of construction project managers. Transactions of Construction Management. 2005;**12**:207-218
- [8] Coesmans P et al. Individual Competence Baseline for Project, Program & Portfolio Management, Version 4.0. International Project Management Association. 2015. ISBN(pdf): 978-94-92338-01-3

- [9] Iyota M, Natori K, Yoshida J, Koike M, Katoh M, Haga T. Practice of a four faculties joint project-based learning for first grade students. Proceedings of Annual Conference of Japanese Society for Engineering Education. 2016. pp. 112-113 (in Japanese). DOI: 10.20549/jseeja.2016.0_112
- [10] Natori K, Kobayashi H, Yoshimura T, Kumamoto K, Kawata Y, Katoh M. Practice of a four faculties joint project-based learning for first grade students (2nd Report). Proceedings of Annual Conference of Japanese Society for Engineering Education. 2017. pp. 464-465 (in Japanese)
- [11] Communication and Interactions. 2017. Available from: https://www.quora.com/Whatis-the-difference-between-interaction-communication-conversation-and-discussion
- [12] Bullinger H-J, Ziegler J. Human-computer interaction: Communication, cooperation, and application design, Lawrence Erlbaum Associates, 1999. In: Mccabe PT, Hanson MA, Robertson SA, editors. Contemporary Ergonomics 2000. 2003. p. 177. DOI: https://books. google.co.jp/books?isbn=0203305361
- [13] Katoh M, Imura N, Washio K. Imperative improvement from failure mode effect analysis to action mode effect, inference, and yields analysis of miss-over-orders. In: Proceedings of SICE Annual Conference. 2017. pp. 1127-1132
- [14] Azuma S, Nagahara M, Ishii H, Hayashi N, Sakurama K, Hatanaka T. Control of Multiagent Systems. Tokyo, Japan: Corona Publishing Co. Ltd; 2015 (in Japanese). ISBN: 9784339033229
- [15] California Polytechnic State University, San Luis Obispo. Multicultural Engineering Program. 2017. Available from: https://mep.calpoly.edu/
- [16] Open Innovation. 2017. Available from: http://www.openinnovation.eu/open-innova tion/
- [17] Kamba H. Global competencies and project manager training. Journal of JSEE. 2013;61(5):5.4-5.9 (in Japanese)
- [18] Kato T. Possibilities of PBL using project management education. Journal of JSEE. 2013;61(5):5.46-5.50 (in Japanese)

Labour Market Competences

Project Management and Learning: The Learning Project

Alfonso J. Gil and Mara Mataveli

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.72051

Abstract

Learning is the key to project management, for the reason that much of the success of a project lies in adapting to the changing environment of any project. The key to change lies not in individual learning, but in the learning of the group that integrates the project, what would be called project learning. The objective of this work is to propose a learning project, a project that can adapt to the changing environments that are generated in the project management. Following the proposals of the so-called learning organizations, a "learning project" model is proposed, which is based on four key components: project leadership, project culture, project structure, and related to the learning opportunities with the project (training for the project). The work ends with a few recommendations toward the management of human resources in project management.

Keywords: organizational learning, management learning, project learning, culture, leadership, training

1. Introduction

Internationalization, the changing needs of customers, intense competition and technological changes, and, consequently, uncertainty and complexity are outstanding environmental characteristics that are currently faced by project managers. To remain competitive and viable, projects need to adjust to implement changes continually. For [1], there is an organizational need to manage projects correctly, as well as learning from successes and failures, capturing, disseminating, and applying lessons learned in project development, in other words, improving organizational learning in project management. References to organizational learning derive from the strategy theory and refer to the ability an organization has to renew its competencies as a dynamic capability [2]. Organizational learning has also been identified as one



© 2018 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. of many factors of success in project management [3, 4]. Besides, organizational learning is a necessary tool for managing uncertainty in projects [5] and may allow for the accumulation and preservation of experience in project implementation [6].

However, organizational learning in projects does not occur regularly, and when it does, the expected results are not always achieved [7, 8]. The focus of learning is on individual learning but does not lead to project learning.

This work aims at proposing a project model that improves learning. It is intended to indicate some key points that serve as a reference for the development of projects and to lay a foundation from which to generate optimal projects, where people can develop their potential for the success of the project. It is an ideal project and a reference model for managers who intend to make learning an essential element in project management. In other words, it is about moving towards a "learning project". A project learns when the performance of the tasks that its members execute individually or collectively improves continuously, either because they improve the procedures or for a greater use of the resources concerning the objectives of the project, but, also, when this improvement accumulates in the "project memory" to be transferred to other projects or works. For this situation to consolidate, there must be a series of premises or foundations that ensure a sustainable learning such as there is a culture toward learning, there is an atmosphere of cooperation between people, the leadership of change and learning is valued and developed, and there is a comprehensive information system.

This paper addresses first a general reflection on orientation about learning in projects; second, group learning, which can help to recognize how learning occurs in the projects; third, a learning project model that is proposed; and four, the approached work contributions to project management.

2. Learning orientation in project management

Learning orientation has been seen as a value of the organization, which affects not only the organizational development but also the creation and knowledge transfer and the innovation [9, 10]. Strategic management academics consider the orientation of learning as a distinctive type of strategic direction and management philosophy that is viewed as a process in which knowledge and information are communicated and disseminated throughout the organization, which helps employees to improve their skills and abilities [11]. More specifically, the learning orientation toward the learning of members of a team refers to the group members' propensity to focus on learning, acquire new skills, master new situations, and develop competencies [12, 13]. Individuals with a dispositional learning orientation focuses on distal and general purposes, and goal setting emphasizes the objectives of a task at a particular level [15]. The general learning orientation focuses on a task at a given level.

In the area of project management, it has been pointed out that, although the content of a project may have a temporary characteristic, the organization of the project is the source and the result of learning [16]. In fact, it has been emphasized that learning helps organizations to adapt their governance structures to maximize the performance of their projects [17], especially if it is taken into account that the transfer of learning is of great importance for any organization that aims at improving its capacity to implement interdisciplinary projects [18].

Also, two aspects related to learning orientation and management of project teams have been identified. First, time pressure has been positively associated with the orientation of learning through intrinsic motivation [19]. Ref. [20] indicated that team members could often see the need to do something quickly or make many things happen simultaneously. Under these circumstances, moderate levels of time pressure could be endogenous to the team members' project, which would lead them to feel positively challenged and to be more involved in learning within the project. Second, learning orientation has been associated with creativity through intrinsic motivation as one of the factors that precisely trigger creativity [21, 22]. In fact, learning orientation plays the role of a motivational process through which intrinsically motivated team members can engage in learning activities, resulting in creative outcomes [23]. Therefore, the effect that time pressure has on creativity is assumed to be positively mediated by learning orientation.

3. Team learning and project management: the project learning

Project management involves people who constitute teams or work groups, so before working on learning in project management, reference is made to teamwork and team or group learning because it is a reference for learning development in project management [24].

Working groups or teams have been defined in a number of ways, following [25] as being: "A team is a collection of individuals who are interdependent in their tasks, who share responsibility for outcomes, who see themselves and who are seen by others as an intact social entity embedded in one or more larger social systems (p. 241)." Team learning is often conceptualized as a continuous process of action and reflection [26] through which teams acquire, combine, and apply knowledge [27]. This process is closely related to activities such as asking questions, seeking feedback, improvising, discussing errors, challenging underlying assumptions, and reflecting on specific outcomes or unexpected results [28]. Team learning, however, also refers to the knowledge that is encoded and embedded in the computer [29], for example, underlining the importance of documenting and reflecting on what has been learned through a process of knowledge codification. Codification involves recording the procedures of teamwork and hence converting tacit knowledge into explicit knowledge so that results can be developed. Repository and retrieval allow learning to persist over time, preserving knowledge and artifacts gained in memory repositories and facilitating team members to find and access them for future use [30].

In the work teams, the importance of team learning has been emphasized, since this learning constitutes a fundamental variable in the effectiveness of the work teams [31, 32].

Also, team or group learning has been identified in different ways by researchers. Authors such as [26, 28, 29] have focused on learning as a process, while others [33] have highlighted learning outcomes. However, group learning could be identified as "an emergent property of the group exerting influence beyond the individual members involved in the original learning process, p. 1043" [34].

The problem that arises is to know how the process of learning in a team is facilitated. In other words, what strategies can be developed to achieve effective group learning and that this is quick and efficient [35]. Building the agile team learning model (ATLM) based on fast task mining, to develop the learning method, shortens the learning period and improves the precision of knowledge acquisition. ATLM divides the whole process of learning into three phases: (1) information acquisition, (2) fast task mining, and (3) team learning.

- 1. Information acquisitions (IA). The objective of this phase is to acquire more relevant information for the work team. In this sense, all members of the team can collect information from different sources of the company, through experts, through network services, or through training courses.
- **2.** Fast task mining (FTM). Building a shared vision is the third discipline of learning organizations [36]. It is about making the shared learning tasks attractive. It is about activating the team members to get a constant process, instead of a passive dedication and little follow-up.
- **3.** Team learning (TL). As soon as the shared learning task has been built, the TL phase should be started immediately to achieve the learning task quickly and accurately. And so, four crucial steps are proposed.
 - **a.** Use case. Create according to real projects or members' interests. Each use case focuses on describing how to achieve the knowledge points in learning the task.
 - **b.** Practice. Team members should play different roles in each case and try to practice use case through teamwork.
 - c. Sharing. Knowledge exchange is a major factor in the development of group learning.
 - d. Test. Team members should share achievements and complement skills for each other.

For all these processes to take place, the work of leadership is critical in the development of team learning. Literature has pointed out that leaders can facilitate both individual learning [37, 38] and team learning [39]. In this sense, [28] showed that leaders who train team members and help solve problems influence group norms, encourage team communication, and improve team learning. Also, [40] found that experience in leaders' experience working with teams, solving problems, and challenging group members improves group learning.

Therefore, the key to learning the project lies in the group learning, it is the group contibutes to the development of the project, which means that the group learning in the project corresponds to the "project learning".

4. Generating a "learning project"

Researchers from large institutions (such as universities or national research institutes) will often move from one project to another, creating opportunities for transfer of learning between projects [41]. The problem that arises is not to make learning dependent only on the intervention of individual people but on the actual exercise of the project. The challenge here is to avoid making learning dependent only on the response of certain people, and rather to focus on the real performance of the project. It is about generating "learning projects." A learning project can be defined as a project that is continuously transformed to adapt to changing contexts and move toward opportunities for improvement.

Love et al. [42] argue that to facilitate learning in project-based environments effectively, three conditions must be established. The first one is to have a clear understanding of the unique characteristics of the project and the operational environment in which it is developed. Second, continuous learning must be integrated into the organizational culture and, third, synergies are created through the collective actions of the agents involved in the project in support of learning. All this supports the idea of moving toward learning projects.

In addition to the focus on learning organizations [43–46], but with the uniqueness of project management [42], four key aspects can be proposed for the creation and development of learning projects: leadership project, culture project, structure project, and learning opportunities project (see **Figure 1**).

4.1. Leadership in the learning project

As [47] point out, projects are a form of temporary organization in which the project members (usually staff members) are supported by their affiliated organizations (usual enterprises), which convey specific project-oriented tasks.

The relationship follower-leader can last for a limited period. Besides, project managers feel that there is an "authority gap" while managing and directing project teams, considering that their subordinates perform multiple roles outside the project in question [48]. However, the task of a leader is more meaningful in projects being carried out in a company that is not temporary, that has the purpose of permanence, since he or she becomes a key figure in



Figure 1. Project as a learning project. Source: Own elaboration.

aligning the members of the project team as to their tasks, which are temporary according to their nature. The problem that arises is to recognize the style of leadership that is more in line with the interests of the project. In general terms, it has been pointed out that person-oriented leadership styles are positively related to team learning behaviors [49].

Thus, it could be said that the leadership style, as a transformational one, could be a better fit in the administration of projects. Other leadership styles such as ethical leadership and values leadership have been identified as appropriate for the management of learning in groups and learning organizations, respectively. For this reason, they are considered suitable for project management, since they represent models through which the agents involved in the project become empowered and become an accurate action guide for the people led.

Research on transformational leadership has attracted considerable academic attention over the past two decades [50–54]. Transformational leadership has been defined as "charismatic, visionary, and inspirational actions that influence followers to broaden their goals and perform beyond the expectations specified in their formal work roles, and job descriptions, p. 286" [52]. Transformational leadership contributes to support behaviors to the team within organizations and acts to improve employee performance, including attitude, behavior, and performance [55–58].

However, leadership in temporary projects may have a different meaning in permanent and temporary organizations. In this sense, transformational leadership, which has been shown to be an effective approach to stimulate high-level commitments in organizations [59], has also been proven to be useful in the context of the project [47].

Ethical leadership has gained attention in recent years, especially the importance of reducing behaviors associated with the lack of ethical conduct from companies and professionals [60]. Interest in ethical leadership comes from the proposition that ethical leadership increases employees' ability to cope with conflict and reduces employee misconduct [61]. Ethical leadership has been positively linked to other behavioral attitudes such as job satisfaction, employee engagement, identification with the organization, and performance of tasks among other behaviors [62]. Ethical leadership is defined as "the demonstration of normatively appropriate conduct through personal actions and interpersonal relationships, and the promotion of such behavior to followers through two-way communication, reinforcement, and decision-making" (p. 120) [63].

Ethical leadership has been singled out as a leadership style that motivates followers to adopt positive psychological states [64] and engage in beneficial behaviors to the organization [65]. Ethical leadership has been associated with group learning, which as pointed out is closely related to learning in projects.

Works such as those by [66] show that ethical leadership fosters the psychological safety of members of work groups. This behavior encourages the proactive participation of followers in decision-making and their propensity to take risks, practices that are fundamental to behavior group learning.

Group learning behavior is defined as "the extent to which [group] members seek opportunities to develop new skills and knowledge, welcome challenging assignments, are willing to take risks on new ideas, and work on tasks that require considerable skill and knowledge (p. 114)" [67]. This behavior allows a rapid adaptation to the current contexts of organizational changes. Therefore, organizations with a proactive behavior of their work teams are more likely to achieve long-term competitive success [68].

Ethical leaders act as models for their followers by setting clear standards and holding employees accountable to those criteria [63]. Ethical leaders serve as models for their followers by setting clear standards by means of making the employees take the responsibility to go after those criteria. Ethical leadership promotes the behavior of group learning by providing a basis for the development of norms and productive practices of group learning [69]. In this sense, ethical leaders encourage the behavior of group learning by assigning a substantial value to group rules and procedures and by providing members with formal systems and training programs that disseminate actual learning to guide group actions and decisions [60].

Another leadership style that shares perspective with ethical leadership, which has also been identified as suitable for the development of learning organizations, is leadership in values. A development of this type of leadership focused on learning organizations can be followed in the work of [70]. For these authors, the figure of a leader in a learning organization combines two different features: (1) knowledge of oneself and others, which are qualities of emotional intelligence and (2) behavior guided by values, the so-called leadership in values (see **Figure 2**).

At present, it is widely known that there is no consensus on the exact nature of emotional intelligence. Several authors point out different factors that can be included in the concept of emotional intelligence [71–74]. Following [72, 73], emotional intelligence comprises five essential characteristics: (1) emotional awareness, to know one's moods and those of others; (2) self-regulation, for the management of one's attitudes; (3) motivation, to facilitate the achievement of goals; (4) empathy, to be aware of the feelings and needs of others; and (5) social skills, to enable mutual and solidarity in relationships.

Therefore, self-knowledge constitutes an important dimension of emotional intelligence. In short, an emotionally intelligent person, in addition to relating well to others and understanding them (interpersonal skills), knows himself or herself and "relates" well to himself or herself (intrapersonal attributes).



Figure 2. Leadership bases in learning organizations. Source: [66].

The field of emotions, and more specifically to self-knowledge and knowledge of others, has been addressed by leading business leadership experts such as [75–79]. Specifically, [75] who addresses the emotional leadership theme. According to him, the importance of the emotional leader lies in the safety of his or her behaviors (emotional stability). Additionally, the new leader must also be able to create group culture and organizational culture, without overshadowing the ability of each in the collective effort. The foundations of emotional leadership are based on values such as commitment (fostering talent and directing it to the common interest), communication (first of all knowing to listen), integrity (sharing of tensions), etc.

The style of leadership proposed by [80] is based on the leader values: when his or her values are correct, he or she will follow proper actions. The values are references concerning the course of the appropriate action; they are guides for decision-making. The path to leadership has two objectives: (1) an internal one, of personal perfection through practice, and (2) an external one, of social harmony, a result of the leader's behavior.

The first objective, personal perfection, seeks to discover and cultivate its good internal qualities in two different ways: (1) prudent thinking, thinking before acting: the person must be open (see situations from all points of view and without bias), altruistic (doing the right thing without seeking the only personal benefit), and disciplined (assuming one's mistakes); (2) the word prudent: value sincerity first, and (3) conservative actions: be careful with actions. It is about combining simplicity with refinement.

The second objective, social harmony, is reflected in true self-confidence. These are the leader's characteristics: (1) he or she is virtuous: he or she tries to do the right things; (2) he or she grants responsibility to the right people; and (3) he or she identifies himself or herself with the interests and well-being of all.

Specifically, the role of a leader in building learning organizations, and also in the construction of learning projects, is to foster learning environments [81]. The fundamental role of leaders of learning organizations is to promote and coordinate actions, resources, and services that improve the learning conditions of the group members [82].

4.2. The culture of a learning project

The concept of organizational culture has originated in the cultural anthropology and has become popular in the study of organizational behavior and the business management [83]. In general, organizational culture refers to the set of values and beliefs that provide norms of behavior for the members of an organization [84]. Organizational culture can be defined as: "the collection of traditions, values, policies, beliefs, and attitudes that constitute a pervasive context for everything we do and think in an organization (p. 1)" [85].

Empirical evidence suggests that organizational culture plays a significant role in organizational effectiveness [83] and knowledge development [86]. And, culture strongly influences the employees' behavior beyond control systems, procedures, and official authorities [87]. Therefore, the organizational culture becomes a powerful tool to obtain the results that are expected by the administration [88]. The culture of an organization is a junction of different cultural orientations [89], which allows the analysis of the various types of cultures in organizations. In line with these approaches, project management generates a project culture that develops values that drive the project.

In the words of [90], "great projects create a revolutionary project culture. The execution of great projects often requires a different project culture, which can spread to an entire organization (p. 20)." Project culture facilitates learning and drives innovation and change, which would mean that the project culture has traits of a culture of learning and innovation, but also develops other values such as trust and commitment.

The learning culture has been defined as the culture that is oriented to the promotion and facilitation of workers' learning, to their participation and diffusion, and to their contribution to organizational development and performance [91]. The learning culture has a significant role in the organizational learning development and in mechanisms that reinforce learning [92, 93]. In this sense, literature revision about learning strengthens the idea that a learning culture is critical in spreading the lessons learned [94, 95].

An important aspect that characterizes a learning culture in a project is the acceptance of errors as a learning process. Repeated mistakes are a characteristic of organizational life [96] and generally make part of this. Learning from experience within an organization should be public, a recorded activity. The author is strongly in favor of postproject reviews. In general, the learned lessons about tasks that had major problems are recorded at the end of the projects. The "learned lessons" documented from past projects become an unpriced source of knowledge in project management. It is quite remarkable that these failures, the lessons that were learned, are fed into an appropriate system and available for consultation by the team.

In spite of being considered important, sometimes the registration of mistakes and failures are not recorded. According to studies of [97] also confirmed in Ref. [96], the main reason for this kind of inaction is pointed out as a lack of employee time. But, the author adds more reasons such as lack of management support and lack of incentive, resources, and clear guidelines. Nevertheless, it is recognized that the wise record can provide time and save energy, besides optimizing. Also, it is an outstanding and unparalleled opportunity for project managers to learn from mistakes not only from others but also from their mistakes and, from there on, to act upon the lessons learned.

Innovation is of great importance to project management. In fact, [98] point out that a project is a primary source for implementing changes. According to [99], there is a tendency in the project management literature to equate the project with an innovative organization, which makes innovation management a success factor for the project [100]. For all this, an innovative project develops a culture that stimulates commitment and behavior toward innovation. Innovation is a process of transforming opportunities into practical use [101]. In general, companies with a greater capacity for change will be able to respond more efficiently to changes in the environment and achieve a sustainable competitive advantage [102]. For this reason, [103] point out that companies need to improve their innovative culture so that all members seek new products, services, or processes.

Values that characterize a project culture are related to those benefits of learning and innovation but are also distinguished by the trust [5] and commitment [104], especially if one considers the uncertainty that sometimes surrounds the management of projects. Risk arises from the factors associated with the project [105], for example, uncertainty in the level of performance of project members, the quality and reliability of the work undertaken, and the alignment of the objectives of each part.

4.3. Structure of a learning project

The organizational structure represents a set of expectations about whom, what rules and procedures should be followed, how decisions are made, and what control should be used [106]. A mechanistic structure provides formalized prescriptions about how members relate to each other and complete their work. Overall, differences in organizational structure can range from flexible, informal, loose, and decentralized operations to rigid, formal, restricted, and centralized services [107].

The organizational management of the project comprises the set of essential components for the administration of the project [108, 109]. The management of organizational projects differs from the level of organization, as it covers the entire organization, including all groups, operations, and projects. Corporate design is then present in three tiers [110]: organization, management of organizational projects, and projects. In this research, we focus on project management. In this sense, the organization chart of a project refers to how the project organization is designed, what competencies are involved, how the work is shared, and how responsibility is assigned [111].

In general, contingency theory represents an important pillar in the study of organizational design [112]. The fundamental idea of this theory is that activities should be organized differently depending on certain exogenous contingency factors. The scientific concern is limited to adequate parameters of contingency, as pointed out by [113], and some studies emphasize the environmental dimensions as the market and others focus on the technology used. This theory has also been valid for the design of projects as [114] pointed out, for which the dimension design of the project is divided into three subdimensions:

- Strategy and structure, representing the permanent environment that is temporarily linked to an organization,
- Management project and team project, including methodologies, tools, and standards applied to the project, and.
- Culture and social processes, which encompass relationships and interactions between people working in and with projects, as well as relationships between projects.

At the design stage, especially to significant and complex projects, the primary challenge is to coordinate the many specialist technicians involved [115]. The coordination of a project requires the exchange of information between members [116], which sometimes becomes complex, especially if it involves coordinating multidisciplinary teams [117].

Communication is a significant factor that contributes to the success of the project [118], for example, communication is crucial for controlling time and cost [24]. Poor communication among project members is one of the leading causes of complications arising during the process of a project [119–121]. For this reason, communication management is essential in projects. Management of communication in projects involves activities that facilitate the creation, distribution, and understanding of information [122].

4.4. Learning opportunities in a learning project

The term "learning opportunities" has a broad meaning; in fact, it could encompass culture and leadership as well as the learning structure [123]. However, this concept can be limited to learning opportunities related to training and development. We are referring to on-the-job training.

The importance of training is that employees acquire new knowledge and skills, as well as new ways of working and new tools. Training provides opportunities for employees to learn throughout their lives and also to develop a career path. On the other hand, the importance of training is also due to the increase in competitiveness, productivity, quality around the company, and the management of projects. Also, investment in training is one of the most important aspects when a project is intended to grow since training translates into productive human capital and an instrument that generates greater growth and performance for projects, which impact satisfaction of customers.

Training is one of the most valuable human resource tools for improving the productivity of workers, groups, and organizations as a whole, and it has been emphasized as an important means of learning in organizations [124]. Also, during the change processes, companies demand training to adapt their skills to the new needs arising from the processes of change [125]. In general, it has been noticed that training is a tool that facilitates innovation and change initiatives [126].

Investment in training has been considered as being very beneficial for the development of projects [127]; especially the importance of training for project managers has been emphasized [128]. Notably, two reasons why training project managers of projects are important have been highlighted [129]. The first reason is that basic knowledge of a project is quite broad; it covers an extensive set of areas of particular knowledge of project management; the project manager must have an in-depth understanding of industry, technology, and management issues, which are likely to be found in the projects. The second reason is that project discipline is both theoretical and practical, and it is not enough that the project manager has an abstract and conceptual knowledge of methods, tools, and management practices, and the director should also be able to apply this knowledge in complex operating environments. Therefore, education and training of project managers should be considered as a multidimensional and complex process. The study by [130] identified three main areas of intervention through training; the first is the development of critical thinking to manage complexity; the second is the creation of simple parameters to manage projects; and the third is the management of interpersonal skills rather than technical skills.

Given the unique characteristics of a project such as its duration determined in time and a particular context, sometimes it will be necessary to intervene directly and promptly in a learning problem or a new situation of change. In these cases, training must be immediate and adjusted to the problem and context, and this is what [131] call "just-in-time training" (JIT-T):

JIT-T means "as needed" training rather than accumulating an inventory of know-how that is lost over time. JIT-T indicates not only at the right moment but also just enough training, and in just the right context. JIT-T may also be considered as a rediscovery of on-the-job training in a self-paced manner [131].

Therefore, training and education have traditionally been considered as fundamental elements of project management [132]. Thus, research shows that training is a key factor in the implementation of projects [133].

5. Conclusion

5.1. Some references of human resource management in project management

Whether developing a system-wide strategic plan, promulgating an organizational development (OD) intervention, producing a new training curriculum, or supporting individual learning at work, human resource development (HRD) activities are more often organized in projects [134]. Project work has become increasingly prevalent in organizations around the world and is an important consideration for the success of the organization [135]. Project management has become an essential organizational competence [136]. As a project management research, the methodology and development of the theory have increased, so that corporate and investment in knowledge and project management personnel have improved [137].

"Human capital is of unique importance in today's economy because a growing number of organizations rely on their employee's knowledge-related capabilities versus their physical labor to achieve business goals (p. 4)" [138]. And undoubtedly, one must agree with the author that this is especially applicable and valid for project teams. As [139] point out, the success or failure of projects depends on the people who are involved in it, so human capital in project management is an important issue to analyze in the management of the projects [140].

The purpose of human resources in the project management area is to build a cohesive team to ensure that the project has the sufficient and suitable personnel, with appropriate skills and experience to fully contribute to the achievement and success of the project.

Some questions must be answered in the election of human resources in a project such as which sorts of resources are required, which number of resources is adequate for the tasks and deliverables, or even, the staff required and the time necessary for phase, what kind of skills are needed, how they will be acquired, and when and for how long they will be required.

Another answer that must be replied to is for instance what core skills, competencies, and experience the professionals need to present to occupy a variety of roles during a project.

Developing a human resource plan, acquiring, developing, and managing a project team are steeped and embedded actions in a project. Developing a project team is the process in which the project manager improves the competencies and team performance as a whole. On the other hand, in specific situations, as faced in developing countries, the shortage of human capital at managerial and executive levels may have unfortunate consequences for successful project completion and [141] have summarized the same in their findings. Management in the context of project management comprises actions such as leadership and team building.

5.2. Contributions of the work to the management of the projects

Following the perspective of the learning organization [36], this work has as main objective to propose a model of the learning project. The idea is that workplaces should encourage and motivate learning so that personal and organizational growth is achieved. But for learning to take place, a series of conditions or factors that influence learning should be facilitated. Among these factors, the following were highlighted: progressive leadership, cooperative teams, openness at work, a realistic workload, mechanisms to recognize staff, and participation in decision-making [142]. This work has developed based on these proposals.

The learning project we propose is based on four pillars related to nontangible aspects of strategy, leadership, culture, structure, and learning facilities. Leadership is the project guide that brings together the shared interests of the project and the leader of a project becomes a leader of learning [143]. Also, through leadership, a culture and a project structure are generated and consolidated. At the same time, structure and culture are interrelated, so an effective communication management plan (as part of the structure of a project) fosters a collaborative culture that in turn develops a cohesive project team [144], which in turn promotes participation in decision-making and lays the groundwork for learning in the project [24]. The last pillar of the model is continuous training. Training forms the basis of the model; thanks to the training, a full range of skills are developed that are necessary for the development of the project, which by its very nature is multidimensional.

In short, the model proposed is based on leadership [145]. Leadership drives a culture and a learning structure, and all this is consolidated through continuous training as a powerful means of learning throughout life.

Also, it would be required to consider that, although project management traditionally fits with a positivist epistemological framework, projects are often characterized as separate (to the "regular" organization) and temporary and the tasks that are performed focus on the unique objectives of the project [146]. However, teams are often socially and contextually intertwined beyond their host organizations and often pursue goals (stated or implicit) that

go beyond traditional measures of real projects such as personal learning, knowledge development, career advancement, or organizational cultural changes [147].

All these aspects should be considered in the development of the project, especially if you take into account that the basis for success or failure of projects depends on the people who are involved in it [139, 140]. The search for a balance between the objectives of the project and the interests of their participants can ensure a better outcome of the project.

Author details

Alfonso J. Gil^{1,2*} and Mara Mataveli¹

*Address all correspondence to: alfonso.gil@unirioja.es

1 Departamento de Economía y Empresa, Universidad de La Rioja, Logroño, Spain

2 Universidad Nacional de Educación a Distancia UNED, Madrid, Spain

References

- Duffield SM, Whitty SJ. Application of the systemic lessons learned knowledge model for organisational learning through projects. International Journal of Project Management. 2016;34(7):1280-1293. DOI: 10.1016/j.ijproman.2016.07.001
- [2] Teece D, Pisano G, Shuen A. Dynamic capabilities and strategic management. Strategic Management Journal. 1997;18(7):509-533. DOI: 10.1002/(SICI)1097-0266(199708)
- [3] Heaton KM, Skok W, Kovela S. Learning lessons from software implementation projects: An exploratory study. Knowledge and Process Management. 2016;**23**(4):293-306
- [4] Cooke-Davies T. The "real" success factors on projects. International Journal of Project Management. 2002;**20**(3):185-190. DOI: 10.1002/kpm.1525
- [5] Atkinson R, Crawford L, Ward S. Fundamental uncertainties in projects and the scope of project management. International Journal of Project Management. 2006;24(8):687-698. DOI: 10.1016/j.ijproman.2006.09.011
- [6] Kerzner H. Applied Project Management. Best Practices on Implementation. New York: John Wiley; 2000
- [7] Milton N. The Lessons Learned Handbook: Practical Approaches to Learning from Experience. Oxford, UK: Chandos Publishing; 2010
- [8] Williams T. How do organisations learn lessons from projects-and do they? IEEE Transactions on Engineering Management. 2008;55:248-266. DOI: 10.1109/TEM.2007. 912920

- [9] Tajeddini K, Altinay L, Ratten V. Service innovativeness and the structuring of organizations: The moderating roles of learning orientation and inter-functional coordination. International Journal of Hospitality Management. 2017;65:100-114. DOI: 10.1016/j. ijhm.2017.06.010
- [10] Sinkula J, Baker W, Noordewier T. A framework for market-based organizational learning: Linking values, knowledge, and behavior. Journal of the Academy of Marketing Science. 1997;25(4):305-318. DOI: 10.1177/0092070397254003
- [11] Duncan RB, Weiss A. Organisational learning: Implications for organizational design. In: Staw B, editor. Research in Organisational Behavior. Greenwich, CT: JAI Press; 1978. pp. 75-123
- [12] Bunderson JS, Sutcliffe KM. Management team learning orientation and business unit performance. Journal of Applied Psychology. 2003;88(3):552-560. DOI: 10.1037/ 0021-9010.88.3.552
- [13] Van de Walle D. Development and validation of a work domain goal orientation instrument. Educational and Psychological Measurement. 1997;57(6):995-1015. DOI: 10.1177/0013164497057006009
- [14] Dweck CS, Legget RL. A social-cognitive approach to motivation and personality. Psychological Review. 1988;95:256-273
- [15] Gong Y, Chang S. The relationships of cross-cultural adjustment with dispositional learning orientation and goal setting. A longitudinal analysis. Journal of Cross-Cultural Psychology. 2007;38(1):19-25. DOI: 10.1177/0022022106295438
- [16] Ahern T, Leavy B, Byrne PJ. Knowledge formation and learning in the management of projects: A problem solving perspective. International Journal of Project Management. 2014;32(8):1423-1431. DOI: 10.1016/j.ijproman.2014.02.004
- [17] Manely K, Chen L. Collaborative learning to improve the governance and performance of infrastructure projects in the construction sector. Journal of Management in Engineering. September, 2017;33(5). DOI: 10.1061/(ASCE)ME.1943-5479.0000545
- [18] Argote L. Organizational learning research: Past, present and future. Management Learning. 2011;42(4):439-446. DOI: 10.1177/1350507611408217
- [19] Baer M, Oldham GR. The curvilinear relation between experienced creative time pressure and creativity: Moderating effects of openness to experience and support for creativity. Journal of Applied Psychology. 2006;91(4):963-970. DOI: 10.1037/0021-9010.91.4.963
- [20] Amabile TM, Mueller JS, Simpson WB, Hadley CN, Kramer SJ, Fleming L, Time pressure and creativity in organizations: a longitudinal field study. 2002; HBS Working Paper # 02-073 (http://www.hbs.edu/faculty/Publication%20Files/02-073_03f1ecea-789d-4ce1b594-e74aa4057e22.pdf)
- [21] Amabile TM. Motivating creativity in organizations: On doing what you love and loving what you do. California Management Review. 1997;40(1):39-58. DOI: 10.2307/41165921

- [22] Gong Y, Huang JC, Farah JL. Employee learning orientation, transformational leadership, and employee creativity: The mediating role of employee creative self-efficacy. Academy of Management Journal. 2009;52(4):765-778
- [23] Dweck KS. Motivational processes affecting learning. American Psychologist. 1986; 41(10):1040-1048. DOI: 10.1037/0003-066X.41.10.1040
- [24] Yap JBH, Abdul-Rahman H, Chen W. Collaborative model: Managing design changes with reusable project experiences through project learning and effective communication. International Journal of Project Management. 2017;35(7):1253-1271. DOI: 10.1016/j. ijproman.2017.04.010
- [25] Cohen SG, Bailey DE. What makes teams work: Group effectiveness research from the shop floor to the executive suite. Journal of Management. 1997;23(3):239-290. DOI: 10.1016/S0149-2063(97)90034-9
- [26] Edmondson AC. The local and variegated nature of learning in organizations: A grouplevel perspective. Organization Science. 2002;13:128-146
- [27] Argote L, Gruenfeld D, Naquin C. Group learning in organizations. In: Turner ME, editor. Groups at Work: Theory and Research. Mahwah, NJ: Lawrence Erlbaum; 2001. pp. 369-409
- [28] Edmondson AC. Psychological safety and learning behavior in work teams. Administrative Science Quarterly. 1999;44:350-383
- [29] Gibson CB, Vermeulen F. A healthy divide: Subgroups as a stimulus for team learning behavior. Administrative Science Quarterly. 2003;48:202-239
- [30] Kostopoulos KC, Spanos YE, Prastacos GP. Structure and function of team learning emergence: A multilevel empirical validation. Journal of Management; 2013. 2013;39(6):1430-1461. DOI: 10.1177/0149206311419366
- [31] Decuyper S, Dochy F, Van den Bossche P. Grasping the dynamic complexity of team learning: An integrative model for effective team learning in organisations. Educational Research Review. 2010;5(2):111-133. DOI: 10.1016/j.edurev.2010.02.002
- [32] Sessa VI, London M. Work Group Learning. New York, London: Lawrence Erlbaum Associates; 2008
- [33] Ellis AP, Bell BS, Ployhart RE, Hollenbeck JR, Ilgen DR. An evaluation of generic teamwork skills training with action teams: Effects on cognitive and skill-based outcomes. Personnel Psychology. 2005;58(3):641-672. DOI: 10.1111/j.1744-6570.2005.00617.x
- [34] Wilson JM, Goodman PS, Cronin MA. Group learning. Academy of Management Review. 2007;32(4):1041-1059
- [35] Yin X, Zhu G, Feng L, Agile Team Learning Model Based on Fast Task Mining. In X. Luo et al., editors. ICWL 2010 Workshops, LNCS 6537. Springer-Verlag, Berlin Heidelberg; 2011, pp. 328-335
- [36] Senge PM. The Fifth Discipline: The Art and Practice of the Learning Organization. New York: Currency/Doubleday Publishers; 1990
- [37] Baer M, Oldham GR, Cummings A. Group learning. Academy of Management Review. The Leadership Quarterly. 2003;14(4/5):569-586
- [38] Krause DE. Influence-based leadership as a determinant of the inclination to innovate and of innovation-related behaviours: An empirical investigation. The Leadership Quarterly. 2004;15(1):79-102. DOI: 10.1016/j.leaqua.2003.12.006
- [39] Raes E, Kyndt E, Decuyper E, Demeyere S, Lismont B, Van den Bossche P, Dochy F. Facilitating team learning through transformational leadership. Instructional Science. 2013;41(2):287-305
- [40] Hirst G, Mann L, Bain P, Pirola-Merlo A, Richver A. Learning to lead: The development and testing of a model of leadership learning. The Leadership Quarterly. 2004;15(3):311-327. DOI: 10.1016/j.leaqua.2004.02.011
- [41] Bark RH, Kragt ME, Robson BJ. Evaluating an interdisciplinary research project: Lessons learned for organisations, researchers, and funders. International Journal of Project Management. 2016;34(8):1449-1459. DOI: 10.1016/j.ijproman.2016.08.004
- [42] Love P, Fong P, Irani Z, editors. Management of Knowledge in Project Environments. Oxford: Elsevier Butterworth-Heinemann; 2005
- [43] Awasthy R, Gupta RK. Is learning orientation in manufacturing and service firms different in India? The Learning Organization. 2011;18(5):392-408. DOI: 10.1108/ 09696471111151738
- [44] Goh S. Improving organizational learning capability: Lessons from two case studies. The Learning Organization. 2003;10(4):216-227. DOI: 10.1108/09696470310476981
- [45] Holton J. Building trust and collaboration in a virtual team. Team Performance Management: An International Journal. 2001;7(3/4):36-47. DOI: 10.1108/13527590110395621
- [46] Randeree E. Structural barriers: Redesigning schools to create learning organizations. International Journal of Educational Management. 2006;20(5):397-404. DOI: 10.1108/0951 3540610676458
- [47] Ding X, Li Q, Zhang H, Sheng Z, Wang Z. Linking transformational leadership and work outcomes in temporary organizations: A social identity approach. International Journal of Project Management. 2017;35:543-556. DOI: 10.1016/j.ijproman.2017.02.005
- [48] Tyssen AK, Wald A, Heidenreich S. Leadership in the context of temporary organizations: A study on the effects of transactional and transformational leadership on followers' commitment in projects. Journal of Leadership & Organizational Studies. 2014;21(4):376-393. DOI: 10.1177/1548051813502086

- [49] Burke CS, Stagl KC, Klein C, Goodwin GF, Salas E, Halpin SM. What type of leadership behaviors are functional in teams? A meta-analysis. Leadership Quarterly. 2006;17(3):288-307. DOI: 10.1016/j.leaqua.2006.02.007
- [50] Dvir T, Eden D, Avolio BJ, Shamir B. Impact of transformational leadership on follower development and performance: A field experiment. Academy of Management Journal. 2002;45(4):735-744
- [51] Lowe KB, Kroeck KG, Sivasubramaniam N. Effectiveness correlates of transformational and transactional leadership: A meta-analytic review of the MLQ literature. Leadership Quarterly. 1996;7(3):385-425. DOI: 10.1016/S1048-9843(96)90027-2
- [52] Qu R, Janssen O, Shi K. Transformational leadership and follower creativity: The mediating role of follower relational identification and the moderating role of leader creativity expectations. Leadership Quarterly. 2015;26(2):286-299. DOI: 10.1016/j.leaqua. 2014.12.004
- [53] Rezvani A, Khosravi P, Dong L. Motivanting users toward continued usage of information systems: Self-determination theory perspective. Computer in Human Behavior. 2017;76:263-275. DOI: 10.1016/j.chb.2017.07.032
- [54] Strom DL, Sears KL, Kelly KM. Work engagement: The roles of organizational justice and leadership style in predicting engagement among employees. Journal of Leadership and Organizational Studies. 2013;21:71-82. DOI: 10.177/1548051813485437
- [55] Avolio BJ, Zhu W, Koh W, Bhatia P. Transformational leadership and organizational commitment: Mediating role of psychological empowerment and moderating role of structural distance. Journal of Organizational Behavior. 2004;25(8):951-968. DOI: 10.1002/job.283
- [56] Bono JE, Judge TA. Self-concordance at work: Toward understanding the motivational effects of transformational leaders. Academy of Management Journal. 2003;46(5):554-571
- [57] Yammarino FJ, Spangler WD, Bass BM. Transformational leadership and performance: A longitudinal investigation. Leadership Quarterly. 1993;4(1):81-102. DOI: 10.1016/1048-9843(93)90005-E
- [58] Zhu Y, Akhtar S. How transformational leadership influences follower helping behavior: The role of trust and prosocial motivation. Journal of Organizational Behavior. 2014;35(3):373-392. DOI: 10.1002/job.1884
- [59] Gundersen G, Hellesoy BT, Raeder S. Leading international project teams: The effectiveness of transformational leadership in dynamic work. Journal of Leadership & Organizational Studies. 2012;19(46):46-57. DOI: 10.1177/1548051811429573
- [60] Walumbwa FO, Hartnell CA, Misati E. Does ethical leadership enhance group learning behavior? Examining the mediating influence of group ethical conduct, justice climate, and peer justice. Journal of Business Research. 2017;72:14-23. DOI: 10.1016/j. jbusres.2016.11.013

- [61] Mayer DM, Kuenzi M, Greenbaum R. Examining the link between ethical leadership and employee misconduct: The mediating role of ethical climate. Journal of Business Ethics. 2010;95(Supplement 1):7-16
- [62] Ng W, Feldman DC. Ethical leadership: Meta-analytic evidence of criterion related and incremental validity. Journal of Applied Psychology. 2015;100(3):948-965. DOI: 10.1037/ a0038246
- [63] Brown ME, Treviño LK, Harrison DA. Ethical leadership: A social learning perspective for construct development and testing. Organizational Behavior and Human Decision Processes. 2005;97(2):117-134. DOI: 10.1016/j.obhdp.2005.03.002
- [64] Avolio BJ, Walumbwa FO, Weber T. Leadership: Current theories, research, and future directions. Annual Review of Psychology. 2009;60:421-449. DOI: 10.1146/annurev. psych.60.110707.163621
- [65] Bedi A, Alpaslan CM, Green S. A meta-analytic review of ethical leadership outcomes and moderators. Journal of Business Ethics. 2016;139(3):517-536. DOI: 10.1007/ s10551-015-2625-1
- [66] Walumbwa FO, Schaubroeck J. Leader personality traits and employee voice behavior: Mediating roles of ethical leadership and work group psychological safety. Journal of Applied Psychology. 2009;94(5):1275-1286. DOI: 10.1037/a0015848
- [67] London M, Polzer JT, Omoregie H. Interpersonal congruence, transactive memory, and feedback processes: An integrative model of group learning. Human Resource Development Review. 2005;4(2):114-135
- [68] Argote L. Organizational Learning: Creating, Retaining, and Transferring Knowledge. Norwall, MA: Kluwer Academic Publishers; 1999
- [69] Cropanzano R, Walumbwa FO. Moral leadership: A short primer of competing perspectives. In: Schminke M, editor. Managerial Ethics: Managing the Psychology of Morality. New York: Psychology Press/Routledge/Taylor & Francis; 2010. pp. 21-52
- [70] Gallego DJ, Gil AJ. La construcción de organizaciones de aprendizaje a través del liderazgo. Universidad & Empresa. 2012;22:43-77
- [71] Bar-on R. Bar-on Emotional Quotient Inventory: Technical Manual. Multi-Health Systems: Nueva York; 1997
- [72] Goleman D. Emotional Intelligence. New York: Bantam; 1995
- [73] Salovey P, Mayer JD. Emotional intelligence. Imagination, Cognition, and Personality. 1990;9(3):185-211. DOI: 10.2190/DUGG-P24E-52WK-6CDG
- [74] Weisinger H. Emotional Intelligence at Work. San Francisco: Jossey-Bass; 1998
- [75] Ruiz M. Liderazgo dinamizador y emocional. Harvard Deusto Business Review. 2004;130: 110-135

- [76] Bennis WG. El fin del liderazgo. Harvard Deusto Business Review. 2001;1:74-81
- [77] Bennis WG. Liderar en momentos de desconcierto. Harvard Business Review. 2002;11:4-11
- [78] Kotter JP. El directivo como líder y como ejecutivo: la simbiosis del éxito. Harvard Deusto Business Review. 2001;1:62-72
- [79] Posner BZ. El líder y la visión de futuro. Havard Deusto Business Review. 2000;99:16-23
- [80] Fernández JA. El código de caballeros: liderazgo basado en valores. Harvard Deusto Business Review. 2005;138:46-55
- [81] Buckler B. A learning process model to achieve continuous improvement and innovation. The Learning Organization. 1996;3(3):31-39. DOI: 10.1108/09696479610119660
- [82] Mumford A. Creating a learning environment. Journal of Professional Human Resource Management. 1996;4:26-30
- [83] Gregory BT, Harris SG, Armenakis AA, Shook CL. Organizational culture and effectiveness: A study of values, attitudes, and organizational outcomes. Journal of Business Research. 2009;62(7):673-679
- [84] Schein EH. Organisational Culture and Leadership. 3rd ed. San Francisco: Jossey-Bass; 2004
- [85] McLean A, Marshall J. Interviewing in organisational cultures. In: Working Paper. University of Bath, Bath, UK; 1993
- [86] Gil AJ, Carrillo FJ. Knowledge transfer and the learning process in Spanish wineries. Knowledge Management Research & Practice. 2016;14(1):60-68
- [87] O'Reilly CA III, Chatman J, Caldwell DF. People and organizational culture: A profile comparison approach to assessing person-organization fit. Academy of Management Journal. 1991;34(3):487-516
- [88] Hogan SJ, Coote LV. Organizational culture, innovation, and performance: A test of Schein's model. Journal of Business Research. 2014;67(8):1609-1621. DOI: 10.1016/j. jbusres.2013.09.007
- [89] McDermott CM, Stock GN. Organizational culture and advanced manufacturing technology implementation. Journal of Operations Management. 1999;17(5):521-533. DOI: 10.1016/S0272-6963(99)00008-X
- [90] Dvir D, Shenhar A. What great projects have in common? MIT Sloan Management Review. 2011;March:18-21
- [91] Rebelo T, Gomes A. Different types of organization, different cultural orientations towards learning: What factors explain this? In: Fanti KA, editor. Applying Psychological Research to Understand and Promote the Well-Being of Clinical and Non-clinical Populations. Athens: ATINER; 2009. pp. 175-186

- [92] Bell E. Organisational culture and learning: A case study. Nurse Education Today. 2016;**33**(11):1337-1341. DOI: 10.1016/j.nedt.2013.02.009
- [93] Eskerod P, Skriver HJ. Organizational culture restraining in-house knowledge transfer between project managers—A case study. Project Management Journal. 2007;38(1): 110-122
- [94] Sense AJ. Learning within project practice: Cognitive styles exposed. International Journal of Project Management. 2007;25(1):33-40. DOI: 10.1016/j.ijproman.2006.06.004
- [95] Von Zedtwitz M. Organizational learning through post-project reviews in R&D. R&D Management. 2002;32(3):255-268
- [96] Busby JS. An assessment of post-project reviews. Project Management Journal. 1999;30(3):23-29
- [97] Rose KH. Cover to cover. [Review of the book post-project reviews to gain effective lessons learned]. Project Management Journal. 2007;38(2):100
- [98] Adler P, Obstfeld M. The role of affect in creative projects and exploratory search. Industrial and Corporate Change. 2007;**16**(1):19-50. DOI: 10.1093/icc/dtl032
- [99] Lenfle S. Exploration and project management. International Journal of Project Management. 2008;26(5):469-478. DOI: 10.1016/j.ijproman.2008.05.017
- [100] Beaume R, Maniak R, Midler C. Crossing innovation and product projects management: A comparative analysis in the automotive industry. International Journal of Project Management. 2009;27(2):166-174. DOI: 10.1016/j.ijproman.2008.09.004
- [101] Tidd J, Bessant J, Pavitt K. Managing Innovation. Chichester: Wiley; 1997
- [102] Calantone RJ, Cavusgil ST, Zhao Y. Learning orientation, firm innovation capability, and firm performance. Industrial Marketing Management. 2002;31(6):515-524. DOI: 10.1016/S0019-8501(01)00203-6
- [103] Škerlavaj M, Song JH, Lee Y. Organizational learning culture, innovative culture, and innovations in South Korean firms. Expert Systems with Applications. 2010;37(9):6390-6403. DOI: 10.1016/j.eswa.2010.02.080
- [104] Meyer JP, Herscovitch L. Commitment at workplace: Toward a general model. Human Resource Management Review. 2001;11(3):299-326. DOI: 10.1016/S1053-4822(00)00053-X
- [105] Ward SC. Requirements for an effective risk management process. Project Management Journal. 1999;September:37-42
- [106] Donaldson L. The normal science of structural contingence theory. In: Clegg SR, Hardy C, Nor WR, editors. Handbook of Organization Studies. Thousand Oaks: Sage; 1996. pp. 57-76

- [107] Burns T, Stalker GM. The Management of Innovation. University of Illinois at Urbana-Champaign's Academy of Entrepreneurial Leadership Historical Reference in Entreprenership; 1994
- [108] Aubry M, Hobbs B, Thuillier D. A new framework for understanding organisational project management through the PMO. International Journal of Project Management. 2007;25(49):328-336. DOI: 10.1016/j.ijproman.2007.01.004
- [109] Sankaran S, Drouin N, Müller R. Cambridge Handbook of Organizational Project Management. Cambridge: Cambridge University Press; 2017
- [110] Aubry M, Lavoie-Tremblay. Rethinking organizational design for managing multiple projects. International Journal of Project Management. 2017. pp. 1-15. DOI: 10.1016/j. ijproman.2017.05.012
- [111] Feldman M, Pentland B. Reconceptualizing organizational routines as a source of flexibility and change. Administrative Science Quarterly. 2003;48(1):94-118. DOI: 10.2307/ 3556620
- [112] Donaldson L. The Contingency Theory of Organizations. Thousand Oaks, CA: Sage; 2001
- [113] Miterev M, Mancini M, Turner R. Towards a design for the project-based organization. International Journal of Project Management. 2017;35(3):479-491. DOI: 10.1016/j. ijproman.2016.12.007
- [114] Hanisch B, Wald A. A project management research framework integrating multiple theoretical perspectives and influencing factors. Project Management Journal. 2011;42(3):2-22. DOI: 10.1002/pmj.20241
- [115] Eriksson T, Kadefors A. Organisational design and development in a large rail tunnel project—Influence of heuristics and mantras. International Journal of Project Management. 2017;35(3):492-503. DOI: 10.1016/j.ijproman.2016.12.006
- [116] Cheung SO, Yiu TW, Lam MC. Interweaving trust and communication with project performance. Journal of Construction Engineering and Management. 2013;139(8):941-950. DOI: 10.1061/(ASCE)CO.1943-7862.0000681
- [117] Senescu RR, Arandal-Mena G, Haymaker JR. Relationships between project complexity and communication. Journal of Management Engineering. 2013;29:183-197. DOI: 10.1061/(ASCE)ME.1943-5479.0000121
- [118] Anantatmula VS. Strategies for enhancing project performance. Journal of Management in Engineering. 2015;31(6):4015013
- [119] Ceric A. Minimizing communication risk in construction: A Delphi study of the key role of project managers. Journal of Civil Engineering and Management. 2014;20(6):829-838. DOI: 10.3846/13923730.2013.802739

- [120] Mahamid I. Micro and macro level of dispute causes in residential building projects: Studies of Saudi Arabia. Journal of King Saud University – Engineering Sciences. 2016;28(1):12-20. DOI: 10.1016/j.jksues.2014.03.002
- [121] Sun M, Meng X. Taxonomy for change causes and effects in construction projects. International Journal of Project Management. 2009;27(6):560-572. DOI: 10.1016/j. ijproman.2008.10.005
- [122] Senaratne S, Ruwanpura M. Communication in construction: A management perspective through case studies in Sri Lanka. Architectural Engineering and Design Management. 2016;12:3-18. DOI: 10.80/17452007.2015.1056721
- [123] Gil AJ, Gallego DJ. La realización de formación continua desde la perspectiva de la organización de aprendizaje. Educare. 2016;52(1):107-126. DOI: 10.5565/rev/educar.701
- [124] Laine E, Gegenfurtner A. Stability or change? Effects of training length and time lag on achievement goal orientations and transfer of training. International Journal of Educational Research. 2013;61(1):71-79. DOI: 10.1016/j.ijer.2013.03.014
- [125] Gil AJ, Garcia-Alcaraz JL, Mataveli M. The training demand in organizational changes processes in the Spanish wine sector. European Journal of Training and Development. 2015;**39**(4):315-331. DOI: 10.1108/EJTD-09-2014-0067
- [126] Carbery R, Garavan TN. Organisational restructuring and downsizing: Issues related to learning, training and employability of survivors. Journal of European Industrial Training. 2005;29(6):488-508. DOI: 10.1108/03090590510610272
- [127] Egginton B. Realising the benefits of investment in project management training: Evidence supporting the need for a more strategic approach. International Journal of Managing Projects in Business. 2012;5(3):508-527. DOI: 10.1108/17538371211235344
- [128] Winter M, Smith C, Morris P, Cicmil S. Directions for future research in project management: The main findings of a UK government funded research network. International Journal of Project Management. 2006;24(8):638-649. DOI: 10.1016/j.ijproman.2006.08.009
- [129] McCreery JK. Assessing the value of a project management simulation training exercise. International Journal of Project Management. 2003;21(4):233-242. DOI: 10.1016/ S0263-7863(02)00026-1
- [130] Ramazani J, Jergeas G. Project managers and the journey from good to great: The benefits of investment in project management training and education. International Journal of Project Management. 2015;33(1):41-52. DOI: https://doi.org/10.1016/j. ijproman.2014.03.012
- [131] Globerson S, Korman A. The use of just-in-time training in a project environment. International Journal of Project Management. 2001;19(5):279-285. DOI: 10.1016/ S0263-7863(00)00012-0

- [132] Finney S, Corbett M. ERP implementation: A compilation and analysis of critical success factors. Business Process Management Journal. 2007;13(3):329-347. DOI: 10.1108/14637150710752272
- [133] Ram J, Corkindale D, ML W. Implementation critical success factor (CSFs) for ERP: Do they contribute to implementation success and post-implementation performance? International Journal of Production Economics. 2013;144(1):157-174. DOI: 10.1016/j. ijpe.2013.01.032
- [134] Canden LL, Egan TM. Human resource development and project management: Key connections. Human Resource Development Review. 2008;7(3):309-338. DOI: 10.1177/1534484308320577
- [135] Packendorff J. Inquiring into the temporary organization: New direction for project management research. Scandinavian Journal of Management. 1995;11:3-16. DOI: 10.1016/0956-5221(95)00018-Q
- [136] Fuller JL. Managing Performance Improvement Projects. San Francisco: Jossey-Bass; 1997
- [137] Kerzner H. Project Management: A Systems Approach to Planning, Scheduling, and Controlling. 7th ed. New York: John Wiley; 2013
- [138] Dickson D, Planning the Human Capital Managing Performance. 2017; Accessed from http://scholarworks.rit.edu/other/714
- [139] Henrie M, Sousa-Poza A. Project management: A cultural literary review. Project Management Journal. 2005;36(2):5-14
- [140] Suhonen M, Paasivaara L. Shared human capital in project management: A systematic review of the literature. Project Management Journal. 2011;42(2):4-16. DOI: 10.1002/ pmj.20211
- [141] Banik A, Bhaumik PK. Project management and development of human capital in the Caribbean: Three case studies. Management Decision. 2006;44(8):1076-1089. DOI: 10.1108/00251740610690621
- [142] Schalk D, Bijl M, Halfens R, Hollands L, Cummings G. Interventions aimed at improving the nursing work environment. Implementation Science. 2010;5(34):1-11. DOI: 10.1186/1748-5908-5-34
- [143] Crick RD, Barr S, Green H, Pedder D. Evaluating the wider outcomes of schools: Complex systems modelling for leadership decisioning. Educational Management Administration & Leadership. 2017;45(4):719-743. DOI: 10.1177/1741143215597233
- [144] Livesey PV. Insights of project managers into the problems in project management. Construction Economics and Building. 2016;16(1):90-103. DOI: 10.5130/AJCEB.v16i1.4600
- [145] Neubert MJ, Hunder EM, Tolentino RC. A servant leader and their stakeholders: When does organizational structure enhance a learder's influence? The Leadership Quarterly. 2016;27:896-910. DOI: 10.1016/j.lequa.2016.05.005

- [146] Sense AJ. The project workplace for organizational learning development. International Journal of Project Management. 2011;**29**(8):986-993. DOI: 10.1016/j.ijproman.2011.01.012
- [147] Sense AJ. The social learning character of projects and project teams. International Journal of Knowledge Management Studies. 2009;3(3/4):195-208. DOI: 10.1504/IJKMS. 2009.028836

Culture and Values Competence in International Investment Projects: HR Management Approach

Alexandr Kokovikhin

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.72183

Abstract

This chapter aimed to identify the impact of labor force diversity on human resource (HR) management (HRM) in multinational enterprises, having investment projects in the countries of Eurasian Economic Union, and to propose recommendations which HR practices are effective in making use of diversity. The chapter shows that institutional theory and resource theory are the theoretical bases of diversity management. In practice, diversity management refers to the 'culture and values' competence, included in the ICB Project Management Knowledge Guide 4.0. Methodology research includes publication research, an in-depth analysis of semi-structure interviews with human resource managers and consultants. Interviews identify approaches and human resource practices in the Russian Federation, Kazakhstan and Belorussia: HR marketing attracts foreign workers mostly under pressure stemming from minority customers and the diversity market. They are predominantly recruited for blue-collar jobs, held rather lower management positions and provided few pro- motion opportunities. Besides, managers have not gotten competencies in diversity management, but they think that it is truly important for managing divers personnel, and they are ready for learning and using diversity management. Findings demonstrate as a business case. Besides, diversity management competences of managers and HR officers are representing as model of competences.

Keywords: culture and values competence, sociocultural diversity, human resource management, diversity management, international investment projects

1. Introduction

There are two main causes of great importance of diversity factor in economic development of Eurasian Economic Union countries such as Russian Federation, Kazakhstan and Belorussia.



© 2018 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. First of all, these countries had sustainable growth of international investment projects from the beginning of the twenty-first century. Second, there is a negative trend for decreasing economically active population. For example, rate of able-bodied population in total population in Russian Federation and such industrial region as Sverdlovsk oblast (4.5 million citizens) decline from 67.6 (64.2) to 64.0 (61.0)%. Unfortunately, it is a long-term trend. On the one hand, this can be explained by falling birth rates as a result of negative economic situation in the early 1990s. Another reason for reducing economically active population is low life expectancy and high incidence of disease, also resulting from the negative effects of transitional reforms (**Figure 1**).

The main human resource management (HRM) problems connected with ethnic and cultural diversity are the following:

- (a) Differences in religious and cultural traditions. Different religions have different potentials for integrating into the Russian, European and American business culture. For example, the Hindus demonstrate a high level of tolerance which is preached in Hinduism and Buddhism. As SIETAR research shows that there are an average of 15% fewer conflicts between Hindu employees and Christian employees than the conflicts within the group consisting of only Christians. At the same time, the number of conflicts between Pakistani employees (Muslims by religion) and Christian employees, on the opposite, is 20% bigger than the average [1].
- (b) Foreign employees' poor knowledge of the official language.
- (c) Inflexible policies of the companies towards ethnic and cultural diversity.

As a result of these problems, potential productivity is not achieved, employee's creativity and innovative activity fall and staff turnover and evasion of performance increase.



Figure 1. Rate of able-bodied population in total population (%).

The experience of solving these problems in the tradition of American and European management is represented by the concepts of managing social and cultural diversity (diversity management) and intercultural management (cross-cultural management). The research in the theory and practice of social and cultural diversity management held in the Sverdlovsk oblast became the basis for this article. The objective is taking into account the experience of the European Union countries to determine the prospects for developing practices of social and cultural diversity management in the organizations of the Sverdlovsk oblast which use foreign labor.

The tasks to be solved to achieve this goal are:

- Identifying the approaches in sociocultural diversity management and the types of managerial competencies required by employers who face sociocultural diversity of staff in the Russian Federation, Kazakhstan and Belorussia.
- Working out proposals and recommendations for the development of social and cultural diversity management practices in the organizations and culture and values competencies [2] of managers in international investment projects.

2. Theoretical basics of using the competence approach in the management of sociocultural diversity (diversity management)

The theoretical basis of social and cultural diversity management (diversity management) is the institutional theory, the resource theory [3] and the concept of 'competence approach' (competency-based management) [4].

Competence approach has a relatively short history of development—about 150 years; however, today it is the basis of the most advanced education systems, for example, the system of European education. Using the concept of competencies in HR management goes back to the early 1980s of the last century and is a response to organizational changes and the move towards higher levels of performance. Today, the use of competency approach in HR management (competency-based management) is a recognized method of management.

Competence-based approach is an approach to the description, assessment and development of the person within which the behavior of the person is considered to be the manifestation of his competences. In foreign scientific literature, there are three main directions of competence-based approach: American (behavioral), English (functional) and integration in France, Germany and Austria. The common things in these directions help to define 'competences' as behavioral characteristics which the individual must have or which he must acquire to effectively cope with the work [5].

Referring to sociocultural diversity management, the competence approach provides methods of evaluation, development, motivation and stimulation of the required competences of employees and managers. Describing the state of research in the field of sociocultural diversity, it should be noted that the sociocultural diversity management is a relatively young industry, known little in Russia and in Eastern Europe. Its development is of great interest, because of the current problem for some ethnic groups to integrate in the Russian society and the issues of interaction with labor migrants, foreign partners in the WTO and foreign investors [6]. The management of sociocultural diversity (diversity management) as a separate direction in the personnel management (human resource management) originated in the USA and was a reaction to a common situation, when the company's staff consists of employees of different ages, nationalities, races, religions, etc. It is based on the principle that differences between people should not prevent their joint work [7].

'Diversity' may relate to various aspects: gender, age, color, religion, cultural differences, etc. In this article the focus is on cultural diversity and its specifics connected with employing foreigners and representatives of ethnic and religious minorities. Sociocultural diversity management is seen by many studies as the key strategic aspect of international companies. For example, in the work of the Canadian scientist of Indian origin Srinivas, it is noted that 'one of the key factors for the success and growth of global companies is that they have a global type of thinking, which includes such skills as curiosity, interest, recognition of the complexity and attention to diversity, the search for new opportunities, belief in progress, continuous improvement, long-term perspective and systematic thinking' [7]. In the work of Barkema, Baum and Mannix, the management of sociocultural diversity is considered as 'challenge of the time, the response to which will help companies to cope with diversity in international business, especially if the companies are able to cope with the undesirable consequences of diversity, such as interpersonal conflicts' [8].

Das and Parker note that there is no single correct or best way to manage sociocultural diversity. The authors prove the hypothesis that there are internal and external factors which define what approach to managing sociocultural diversity will be selected by the company, involved in international business. These authors propose the following typology of approaches to diversity management:

- Resistance-denying diversity and the need to manage it.
- Discrimination and fairness discrimination of specific groups in the name of justice for all.
- Access and legitimacy—access to new markets is complicated without involving employees with different cultural, social and other characteristics; legitimacy refers to the trust that people have towards someone who looks and/or behaves in the same way as they do.
- Learning and effectiveness—forming the attitude to diversity in organizations that allows to incorporate different points of view of employees to improve the efficiency of individual decisions, strategies and rules helps to redefine the mission and change the corporate culture. In this approach, it is recognized that often the choices made by workers and the way they implement these decisions depend on their social and cultural background [9].

In the work of Shtykhno, there is a classification of methods of diversity management, based on the approaches Dass and Parker:

- (a) Sustaining homogeneity
- (b) Assimilating individuals
- (c) Celebrating difference
- (d) Acculturation and pluralism [1]

Thus, the higher the external pressure on the organization is and the higher it prioritizes diversity, the better this organization integrates diversity management into other organizational and production processes. The success of diversity management implementation depends on the corporate culture, business strategy and HR policy, which develop internal communications and affect the exchange of information and creativity. This has a positive impact on the employees' performance and, consequently, the performance of the company.

Kohan, Bezrukova and others' works describe the results of researches done to identify the effects of using the methods of diversity management:

- The efficiency of innovative management is directly proportional to the degree of racial diversity and gender balance.
- Staff development policy, including coaching, open communications and consultations with employees, reduces the negative impact of cultural differences on teamwork.
- In some areas intercultural diversity contributes to the business. Companies giving importance to the integration and training on the diversity management are more successful than homogeneous organizations that do not practice staff upgrade training.
- The effectiveness of the approach depends on the size of the group. In large groups or organizational units, diversity matters more than in small ones.
- Contrary to popular belief, the diversity in the team does not necessarily help to attract diverse customers [10].

Analysing the experience of German companies, Ziegert notes that 'an acute shortage of competencies in the field of intercultural communication is mainly experienced by companies operating internationally' [11]. The role of intercultural competences is considered within a separate concept of management—'intercultural management'. Intercultural management issues refer to those organizations that 'employ foreign workers, as management needs to take into account their motivation system' and 'transnational corporations... which should take into account the cultural specifics of different countries in adapting the management style practiced by the company headquarters' [12]. An ability to effectively interact with different cultures is an important criterion for highly qualified specialists and, as a consequence, the success factor in creating innovation and the growth of economic well-being, entering new markets and preserving the existing markets. Based on the above-mentioned results of the research in the USA, Florida introduces a new formula for economic growth, 'describes it as the so-called "Three Ts": Technology, Talent and Tolerance, which contribute to economic development, market attractiveness and competitiveness' [13].

3. Study of sociocultural diversity management practices in the Russian Federation, Kazakhstan and Belorussia

A study of administrative practices in the organizations of the Russian Federation, Kazakhstan and Belorussia experiencing sociocultural diversity was held in the first half of 2017. The researchers used the methodological approach proposed by Professor Dvořáková [14]. The methods included the accumulation of data and information using multiple methods: analyzing scientific publications and conducting semi-structured interviews and observation. The data included both historical and present-day information about the given social sphere. The reasons for using specified methods are connected with studying the social phenomena: national culture, values, behavior and HR management practices. Thus, the collection, selection and analysis done by using the methods of interpretation (reveal ideas (meanings) generated in human economic activity show the sphere for their implementation) provide the basis for achieving visible results and developing recommendations. These methods are used in the in-depth analysis of data obtained from semi-structured interviews with managers and HR specialists to explore approaches and practices in the field of diversity management in the national cultural environment as in the business model.

The object of the research on diversity management practices in the Russian Federation, Kazakhstan and Belorussia was the 74 organizations involved in the three types of activity that provide employment to more than 80% foreign workers (**Table 2**):

- Manufacturing 14 organizations (number of employees over 1000 people)
- Construction 24 organizations (number of employees over 100 people)
- Wholesale and retail trade—26 organizations (number of employees over 200 people)
- Other community, social and personal services—10 organizations (number of employees over 200 people)

If respondents from Kazakhstan and Belorussia were all countries location, than such from Russian Federation were locate in Sverdlovsk oblast. Sverdlovsk oblast is traditionally an industrial region with diversified economy and developed labor market. Structure and dynamics of Sverdlovsk oblast labor market are such as structure and dynamics of Russian labor market. In terms of ethnic diversity, Sverdlovsk oblast holds one of the leading positions in Russia. There are 160 ethnic groups living in the region. According to the all-Russian census of 2010, the Russians make up the significant majority in the population of the Sverdlovsk oblast (3,684,843 people)-85.7% of all the local population, that is, 90.6% of those people who identified their ethnicity. The second largest ethnic community in the region is the Tatars (143,803 people)-3.3% of all the local population in the region. The third largest ethnic community is the Ukrainians (35,563 people)-0.8% of all the local population in the region. The bashkirs make up the fourth largest community. There are 31,181 representatives of this ethnic group living in the Sverdlovsk oblast. The Mari people take the fifth position; their number is 23,801 people. These are followed by the Azerbaijani, Udmurts, Belarusians, Armenians, Tajiks, Chuvashes, Kirghiz, Mordovians, Jewish, Germans and others.

Three cultures defined the ethnocultural distinctness of the Sverdlovsk oblast—Finno-Ugric (Mansi, Udmurts, Mari, Mordovians), Turkic (Tatars, Bashkirs) and Slavic (Russians, Ukrainians, Belarusians). Besides, Sverdlovsk oblast is one of the multireligious regions of the Russian federation. As for 01.01.2016, there are 770 religious organizations which represent 20 world religious denominations. There are also more than 100 ethnocultural organizations and 22 ethnocultural autonomies. From 2011 to 2014, there was an annual growth in the number of non-resident labor force in the Sverdlovsk oblast. So in 2011, the number of non-resident labor force was 61,757, and in 2014 this figure was 80,169 (an increase of 129.8%). There was such an increase due to the growing number of non-residents working in the Sverdlovsk oblast on the patents (from 16,133 people in 2012 to 56,446 people in 2014) (**Table 1**).

In 2015, the number of patents granted to work decreased to 46,000 (by 28%). But it should be taken into account that in 2015–2016 the previously granted work permits are still valid: some categories of people continue working in Russia on permission. There are also some people who chose not to formalize their employment because of the cost of procedure (13.2 thousand roubles) and problems with law. Taking into account all the above-mentioned categories of people, the Federal Immigration Service Office in the Sverdlovsk oblast registered 313,621 foreign citizens in 2015 (337,955 in 2014). Mostly, these are citizens of Tajikistan, Uzbekistan, Kyrgyzstan, Vietnam and China.

The respondents were 82 line managers and HR managers who filled in questionnaires or took part in semi-structured interviews on the questionnaire. The questionnaire aims to identify HR management practices which are used when the staff consists of foreign workers and native citizens from ethnic and religious minorities. Besides, it measures managerial competences in the field of diversity management and cross-cultural management, as well as understanding the need to acquire (deepen) these competencies (**Table 2**).

The study showed prevailing practices of recruiting citizens of Russia coming from ethnic and religious minorities for jobs that require both unskilled and highly skilled labors. One hundred percent of employers confirmed the use of such practices. None of the respondents mentioned the cases of any conflicts related to sociocultural differences between the citizens of the Russian Federation.

The results of our study are consistent with the results of a study conducted by the centre for the promotion of ethnocultural associations under the Urals State Mining University in 2012. The research theme was 'The attitude of members of ethnic and cultural public associations to the development of ethno-religious relations in the Sverdlovsk oblast'. Of the 388 respondents representing 70 ethnic and cultural associations of the Sverdlovsk oblast, 94% of respondents assessed the level of relations between the representatives of different peoples in the Sverdlovsk oblast as 'satisfactory', 'good' or 'very good' [15].

The results of the studies conducted in the Russian Federation, Kazakhstan and Belorussia differ significantly from the results of the study on diversity management practices organized by the group of Professor Korgova in the republics of the North Caucasus. The study revealed that 80% of organizations work under multinational management teams. However, 92% of CEOs prefer to choose the management team and employees representing one ethnicity [16].

Indicator	All foreign workers and people/share of employment in the economy (%)			Deviation (%)			
	2011	2012	2013	2014	2012-2011	2013-2012	2014–2013
The number of foreign workers residing in the Sverdlovsk oblast	61,757 (2.79)	62,872 (2.93)	63,056 (2.93)	80,169 (3.74)	101.81	100.29	127.14
By types of activity							
Agriculture, hunting and forestry	2545 (2.45)	978 (1.34)	795 (1.42)	1431 (2.12)	38.43	81.29	180.00
Mining	_	-	69 (0.11)	6 (0.01)	_	_	8.70
Manufacturing	6320 (1.26)	4834 (0.99)	5289 (1.11)	4480 (0.93)	76.49	109.41	84.70
Construction	13,640 (10.04)	9168 (6.88)	9148 (6.88)	5298 (3.99)	67.21	99.78	57.91
Wholesale and retail trade; repair of motor vehicles, motorcycles, household goods and personal utensils	13,601 (3.84)	7248 (2.16)	4999 (1.48)	5956 (1.77)	53.29	68.97	119.14
Transport and communications	2647 (1.24)	2140 (1.02)	2381 (1.15)	1920 (0.95)	80.85	111.26	80.64
Financial activities			3 (0.01)	6 (0.02)	_	_	200.0
Education	-	-	17 (0.01)	10 (0.01)	_	_	58.82
Other community, social and personal services	6871 (10.16)	5165 (7.63)	5618 (6.91)	4576 (5.94)	75.17	108.77	81.45
The number of foreign workers engaged in labor activities in the Sverdlovsk oblast on the basis of a patent	16,133	33,339	34,639	56,446	206.65	103.90	162.96
Individuals	16,133	33,339	34,639	56,446	206.65	103.90	162.96
Legal entities	-	_	_	_	_	_	_

Table 1. Dynamics in the number of foreign workers who received work permit or a patent in the Sverdlovsk oblast in

2011–2014.

Meanwhile, the majority of managers and HR specialists (90%) do not consider it necessary to address the special social and cultural competences in managing employees who are the citizens of the Russian Federation representing ethnic and religious minorities. This result may be explained by insignificant differences in labor and consumer behavior of the Russian Federation citizens representing different ethnic and religious groups.

No	Questions
1	Are foreign workers of only one ethnic group or different ethnic groups recruited for unskilled/ skilled/highly skilled jobs?
2	Were special social and cultural competences taken into account when deciding to recruit foreign workers?
3	Are foreign workers' special social and cultural competences considered in your organization's foreign labor management?
4	Were there any cases of conflicts in your organization caused by social and cultural differences of foreign workers?
5	Does your organization provide unskilled/skilled/highly skilled jobs to the Russian Federation citizens representing ethnic and religious minorities?
6	Are special social and cultural competences taken into account in the management of the Russian Federation citizens representing ethnic and religious minorities?
7	Were there any cases of conflicts caused by the social and cultural differences of the Russian Federation citizens representing ethnic and religious minorities?
8	What is the most challenging management problem related to staff diversity? What management functions were carried out to solve this problem?
9	Are social and cultural diversity values part of your organization strategy?
10	What is the willingness of the heads of departments to take effective decisions which take into account sociocultural diversity of the staff?
11	What experience (competencies) have you received, working with the representatives of another culture?
12	Are you familiar with the methods of diversity management and cross-cultural management? What activities did you organize (participated in) to create the competencies and understanding of sociocultural diversity problems?
13	Do you consider it necessary to deepen professional knowledge in the field of diversity management and cross-cultural management?

Table 2. Questions for analyzing the diversity management practices in the Russian Federation, Kazakhstan and Belorussia.

The situation with foreign workers is radically different: 100% of respondents recruit foreign workers for unskilled jobs, only 30% hire foreign workers for skilled jobs and none of the respondents confirmed hiring foreign workers for highly skilled jobs.

Sixty-seven percent of respondents spoke about the past or present conflicts between managers and foreign employees as well as conflicts between foreign workers and nationals of the Russian Federation caused by sociocultural differences of the foreign workers. This result is also consistent with the research data of the centre for the promotion of ethnocultural associations under the Urals State Mining University, in which 58% of respondents assessed the impact of the inflow of migrant workers on ethnic and confessional relations as 'increasing tension' and 'bringing a lot of harm'.

However, 54% of respondents noted taking into account particular social and cultural competencies when deciding to employ foreign workers. The proportion of foreign workers employed in construction was 80%, and the proportion of those employed in sales was 67%. Foreign workers employed in the utilities and manufacturing industries accounted for less than 50%. All respondents agreed on the necessity of and confirmed the fact of taking into account foreign workers' special social and cultural competences in the organization's management practices.

Among the most serious management challenges related to staff diversity, there were organizing internal communications (poor knowledge of the Russian language by foreign workers, lack of work contact with the employees who are the citizens of the Russian Federation), increasing the loyalty of foreign workers. Among the measures being taken, there were organizing trainings of foreign workers in the Russian language and culture (required by the Russian legislation).

Eight respondents (10%) proved to be familiar with diversity management methods and cross-cultural management. However, no special events aimed to create competences and understand sociocultural diversity challenges which were held in the organization. The values of sociocultural diversity were not part of the organization's strategy.

All the respondents noted the need to deepen professional knowledge in the field of diversity management and cross-cultural management.

4. Culture and values competence of managers

Eight respondents, which were competent in diversity management, note that competences are important for managers in organizations with sociocultural diversity of staff (**Table 3**).

No.	Group of competences	Competences	Respondents (%)
1	Managerial competence	Recruitment and selection of staff, taking into account sociocultural competences	100 100
		Controlling and assessing sociocultural diversity of staff	100
		Motivation and stimulation of sociocultural competencies	
2	Basic social competence	Openness and interest in the new	50
	(soft skills)	• Flexibility and respect for other people and other cultures	100 75 75
		• The ability to adjust themselves to the changing circumstances of another culture	75
		• The ability to act successfully in a new cultural context	
3	Communicative competence	• The ability and desire to communicate effectively, taking into account the sociocultural diversity of the staff	100 75
		• Style of communication, taking into account sociocultural diversity of the staff	
4	Knowledge of national culture (hard cultural skills)	Knowledge about the national characteristics	100

Table 3. Culture and values competence of managers in international investment projects.

Results of this researching show that competences in nowadays management of diversity are more important for managers and HR specialists (75–100% of respondents) than competences in development of diversity (50–75% of respondents).

5. Recommendations for diversity management in the organization

Based on the labor legislation, experience of managing sociocultural diversity in leading multinational companies and research into diversity management practices in the Sverdlovsk region, we recommend taking into account the following measures to develop diversity management practices:

- To include the sociocultural diversity management in the HR policy of the organization (Figure 2)
- To conduct an audit of HR policies and processes to avoid discriminatory conditions
- To collect and analyze information about the sociocultural environment in the organization
- To develop the policies and techniques of diversity management (including them in the planning, recruitment, selection, training and development of the staff and in the performance management, motivation and stimulation, wage and labor relations)
- To organize staff training in the issues of sociocultural diversity
- To provide information support to the initiatives in the field of diversity management



Figure 2. Sociocultural diversity in HR management system.

- To increase the involvement of employees representing ethnocultural minorities, through special training and organizing intra-network of communications
- To organize the popularization of diversity management practices using informal internal communications

6. Conclusion

The desire to respect and to strengthen cultural diversity in the company is motivated not only by ethical but also purely rational considerations, a direct count on success. It comes to those companies who understand that competitive challenge in dynamic global markets requiring constant innovation and creativity, organizational flexibility and adaptive capacity should better be responded by combining the experience and knowledge of the multinational team.

Such a strategy becomes a powerful motivating tool that can greatly facilitate the implementation of the diversity management system in the enterprise. The task of management is to create the framework conditions in which the enterprise could live and work on the principles of respect for human dignity and respect for human rights and recognition of people of a different culture as equal partners. Diversity management requires, on the one hand, to recognize the value of the 'diversity' phenomenon and, on the other hand, to be prepared for 'cultural transformation', for creating and strengthening the corporate culture which stimulates the differences.

A study of management practices in organizations of the Sverdlovsk oblast showed the absence of conflicts related to sociocultural differences between citizens of the Russian Federation representing ethnic and religious minorities. None of the respondents spoke to them. At the same time, the study revealed conflicts between managers and foreign employees, as well as between foreign workers and workers who are citizens of the Russian Federation. All the respondents agreed on the necessity of and the fact of taking into account special social and cultural competences of foreign workers when developing management practices in the organization.

Among the most serious management challenges related to staff diversity, the respondents identified organizing internal communication (foreign workers' poor knowledge of the Russian language, lack of worker contact with the employees who are the citizens of the Russian Federation), increasing the loyalty of foreign workers.

A study of management practices in socially and culturally diverse organizations of the Sverdlovsk oblast revealed the managers' and specialists' lack of knowledge in the field of diversity management. At the same time, managers and professionals realize the importance of the problems to be solved using the methods of diversity management and the need to deepen the knowledge in this area and put it into practice.

The diversity management within the framework of traditional HR management processes provides a significant economic effect. Thus, an important objective of diversity management is developing the methods of HR management based on the use of staff diversity. The use of these methods in practice makes the staff diversity management one of the directions of the organization's HR policy. Further research in the field of diversity management requires the surveys of foreign workers' opinions on the management methods. It requires building the models and profiles of sociocultural competences of managers and specialists as well as in-depth studying the specific processes of managing socially and culturally diverse staff.

Author details

Alexandr Kokovikhin

Address all correspondence to: gov66@inbox.ru

Ural State University of Economics, Russian Federation

References

- [1] Shtihno DA. Diversity management and business strategy. Human Capital and Vocational Education. 2013;3(7):48-52
- [2] Coesmans P, Fuster M, Shreiner JG, Goncalves M, Huynink S, Jaques T, Pugacevskis V, Sedlmayer G, Thyssen D, Tovb A, Vukomanovich M, Young M, editors. Individual Competence Baseline for Project, Program and Portfolio Management, Version 4.0. 1st ed. Zurich, Switzerland: International Project Management Association (IPMA); 2015. p. 431
- [3] Shen J, Chanda A, D'Netto B, Monga M. Managing diversity through human resource management: An international perspective and conceptual framework. The International Journal of Human Resource Management. 2009;**20**(2):235-251
- [4] Deist le DFO, Winterton J. What is competence? Human Resource Development International. 2005;8(1):27-46
- [5] Kokovikhin A, Sharapova N, Borisov I. National system of certification of worker's competence and qualification as the institute of labour market. The 10th International Days of Statistics and Economics. Conference Proceedings. Prague, Czech Republic, September 8-10, 2016; Prague: VSE; 2016. pp. 871-881
- [6] Kokovikhin AYu, Ogorodnikova ES, Williams D, Plakhin AYe. Institutional Factors in the Evaluation by the Entrepreneur of Municipality Investment Climate, Ekonomika Regiona-Economy of Region. 2017;13:80-92
- [7] Srinivas KM. Globalization of business and the third world: Challenge of expanding the mindsets. Journal of Management Development. 1995;14(3):26-49
- [8] Barkema HG, Baum JAC, Mannix EA. Management challenges in a new time. Academy of Management Journal. 2002;45(5):916-931
- [9] Dass P, Parker B. Strategies for managing human diversity: From resistance to learning. Academy of Management Executive. 1999;**13**:68-80

- [10] Kochan Th, Bezrukova K, Ely R, Jackson S, Joshi A, Jehn K, Leonard J, Levine D, Thomas D. The Effects of Diversity on Business Performance: Report of the Diversity Research Network, Cambridge; 2002
- [11] Zigert A. Trans-migration of high-skilled personnel. Foresight. 2008;3:10-21
- [12] Kumar BN. Interkulturelles Management: Handbuch der Unternehmensfuhrung. Wiesbaden. 1995;S. 684
- [13] Thomas RR. Beyond race and gender. Unleashing the Power of your Total Work Force by Managing Diversity. New York; 1991. p. 410
- [14] Dvorakova Z. Demographic trends and diversity management: the case of the CR. The 10th International Days of Statistics and Economics. Conference Proceedings. Prague, Czech Republic, September 8-10, 2016; Prague: VSE; 2016, pp. 345-350
- [15] Information on: http://minsport.midural.ru/tmp_file/file_5152c78b46445.pdf
- [16] Korgova M. The problems of multinational commands in firms and the ways of decisions. Human Resources Management. 2004;10:20-31

Key Competences for Employment

Traceability of Intra- and Interpersonal Skills: From Education to Labor Market

Alberto Cerezo-Narváez, María José Bastante Ceca and José Luis Yagüe Blanco

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.71275

Abstract

Both educators and employers agree there is a growing gap between competences that labor market expects from its new employees and skills they own. Literature review holds that a set of nontechnical, professional abilities and intra- and interpersonal attitudes are required to close this gap and indicates that more training of soft skills is needed to access employment and success in work life. Although these skills are theoretically included in educational stages, project management approach can be incorporated to improve students and new employees' practical curricula. The methodology consists of the critical review of the competency frameworks established by the DeSeCo and Tuning projects, confronting them against the requirements currently demanded by labor market, based on the reports of Deloitte, PwC, EY, and KPMG, to detect inconsistencies between educational and professional stages and check if project management standards, by PMI and IPMA, cover them. Compiling these weaknesses, actions can be established aimed at solving them, based on project management proposals. The incorporation of project management concepts into educational stages, especially the vision by competences, contributes to improve the employability by highlighting those transverse but essential skills that lead to versatile and successful professionals. To achieve this, it is necessary to care for human competences.

Keywords: intrapersonal competences, interpersonal competences, soft skills, twenty-first century skills, DeSeCo project, Tuning project, IPMA ICB, PMI PMCDF

1. Introduction

In the knowledge society, as van Laar et al. expose [1], organizations operate in a global economy characterized by an intense competition, interdependence, and collaboration.



© 2018 The Author(s). Licensee InTech. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. For Sliter [2], economic, demographic, and technological changes have ushered in a revolution of globalization and rapid innovation, needing a method of describing requirements to accommodate this unpredictability.

Besides, as Neubert et al. hold [3], in modern organizational work environments, the classic career approach has broadly been replaced by new paradigms that minimize organizational factors and stress the importance of an individual set of skills, including transversal ones.

In this environment, as Rodriguez et al. argue [4], competency modeling's inherent adaptability allows it to easily adapt complex, changeable positions, and the nonroutine and interactive tasks required therein. For Bonilla [5], the competence-based education (CBE) has multiple applications for the development of people, organizations, and society, as a whole, highlighting the link between education and labor.

Education programs aim to prepare students for the workplace, as Rainsbury et al. conclude [6], by developing generic and specific competencies useful to students and employers.

2. Scope

This chapter studies the traceability of intra- and interpersonal skills demanded by labor market from educational stages, investigating how project management by competencies approach can help to correct the gaps detected through continuous formation programs, ready to success into the labor market in a dynamic and changing context.

The research is focused on the Latin America countries, Portugal, and Spain, from their regulatory frameworks to practical research studies.

2.1. Objectives

The main objective is to establish a traceable sequence of every intra- and interpersonal skill demanded by the workplace, checking if it is legally collected during learning phases, developed in training and properly applied to working life, like **Figure 1** shows. Likewise, other goals are:

- To contrast that focusing on competency project management approaches is an effective method to implement transversal skills in students and new employees, improving their satisfaction, productivity, and efficiency
- To establish a theoretical framework of intra- and interpersonal skills, which should be taken into account in order to currently succeed into the labor market, ready to be collated, by statistical study and/or case study in future research.

2.2. Methodology

The methodology consists of the critical review of the competency frameworks established by scientific literature, from three points of view: education, project management, and workplace.

Traceability of Intra- and Interpersonal Skills: From Education to Labor Market 89 http://dx.doi.org/10.5772/intechopen.71275



Figure 1. Research framework.

In educational stages, Definition and Selection of Competencies (DeSeCo) Project of the Organization for Economic Co-operation and Development (OECD), during the preuniversity period, and the Tuning project for the European Higher Education Area (EHEA) and América Latina Formación Académica (ALFA), at the university stage, offer and develop the legal framework in which stakeholders have to act. These approaches will be compared and contrasted to topic researches.

Project management by competencies, in this investigation, has a double meaning. On the one hand, it is an emerging profession and, on the other, thanks to its intrinsic transverse and humanistic condition, it covers management requirements in any sector. From the analysis of standards, baselines, knowledge bodies, and frameworks of the two oldest organizations in project management, both the International Project Management Association (IPMA) and the Project Management Institute (PMI), intra- and interpersonal skills and competences are collected and prioritized according to topic researches.

Based on the reports of the most prestigious consulting firms, Deloitte, PricewaterhouseCoopers (PwC), Ernst & Young (EY), and KPMG, the intra- and interpersonal skills most demanded by labor market are filtered and endorsed by topic researches.

With this method, inconsistencies between educational and professional stages are detected, so that improvement actions can be proposed from project management.

3. State of the art

Cleary et al. define employability skills as generic capabilities, key skills which play a significant role in contributing an individual's effective and successful participation in the workplace [7].

In this context, for Gibb [8], generic skills are essential for employment and personal development, fulfillment, community life, and active citizenship. These skills must own, in order to be useful, these characteristics:

- Preparing for employment
- Emerging at entry levels within industry, thanks to be generic
- Equipping individuals to participate effectively in workplaces and adult life
- Being able to be learned
- Being amenable to credible assessment

3.1. Concept of intra- and interpersonal skills

Hard skills can be defined as the specific knowledge and abilities required for a job. However, soft skills are much more difficult to define and measure. They are the interpersonal (people, social) skills that help oneself to successfully interact with others in the workplace, and interpersonal (human, individual) skills that allow oneself to know, understand, and manage own cognition and emotions. In **Table 1**, some definitions from literature are presented:

In summary, soft skills are social abilities and individual attributes, which can also be called as twenty-first century skills by many authors, such as Kraiger [14], Gibb [8], Nealy [15], Gewertz [16], Ziegenfuss [17], Hodge and Lear [18], Suto [19], Soland et al. [20], Hayes [21], Davis [22], Neubert et al. [3], Su et al. [13], Schooner et al. [23], or Ali et al. [24].

For the Oxford Dictionary, Personal attributes that enable someone to interact effectively and harmoniously with other people.

For Rainsbury et al. Behavioral skills required for the application of hard skills and knowledge in organizations.

For Perreault,

Personal qualities, attributes, or the level of commitment that allow an individual to distinguish from others who may have similar skills and experience.

For James and James,

Set of abilities or talents that an individual can bring to the workplace.

For Bonilla,

Application of practical knowledge through physical and intellectual skills and abilities, with respect to criteria or standards of expected performance.

For Su, Golubovich, and Robbins,

Group of constructs that are used to refer to important predictors of readiness and success in the workplace across multiple domains of individual differences and beyond the knowledge acquired in formal education.

Extract of [5, 6, 9-13].

Table 1. Definition of soft skills.

For the Collins English Dictionary,

Desirable qualities for certain forms of employment that do not depend on acquired knowledge, among are included the common sense, the ability to deal with people, and a positive flexible attitude.

3.2. Soft skills in educational stages

In the educational context, Cullen defines competence as complex integrated capacities, in different degrees, in which education must train individuals so that they can operate as responsible subjects in different situations and contexts of their social and personal life, knowing how to see, do, act, and enjoy properly, assessing alternatives, choosing appropriate strategies, and taking responsibility for the decisions taken [25].

Besides, as Ali et al. highlight [24], education requires the integration of relevant content, skills, and instructional support to enhance knowledge processes in line with twenty-first century employability requirements. However, authors, such as Gewertz [16], Bronson [26], Klaus [27], Mitchell et al. [28], or Tan et al. [29], among others, conclude that current students (future employees) do not have the set of soft skills they need to be successful in workplace.

In fact, the lack of soft skills may truncate promising careers with technical abilities and professional expertise but with no human qualities like Klaus affirms [27]. Likewise, as Nabi and Bagley expose [30], recent graduates tend to rate the importance of generic transferable skills more highly than their own ability in those ones.

Even the International Youth Foundation (IYF) publishes the gap among students' characteristics and employers' demands is increasing, due to the lack of soft skills, which is avoiding the achievement and success of the majority of entry-level candidates [31].

Soft-skills education has been overshadowed by the quantitative focus in most educative programs, despite the weak relationship found between curricula and career success, as Laud and Johnson asseverate [32]. Likewise, as Hassan et al. conclude [33], soft skills have been very difficult to embed in teaching and learning courses.

However, as Low et al., Vázquez and Liesa discuss [34, 35], through a review of academic programs, it is possible to improve the acquisition of these transversal skills students are going to need in their professional and personal lives. In this regard, Tito and Serrano remark the importance that universities prepare students in soft skills and graduate with tools that add a differentiating advantage that allows them to compete in the labor market and generally maintain a high sense of well-being with themselves [36].

3.2.1. DeSeCo project for preuniversity stage

The DeSeCo project of the OECD defines competencies as the abilities to successfully meet complex demands in a particular context, implying the mobilization of knowledge, cognitive, and practical skills, as well as social and behavior components such as attitudes, emotions, and values and motivations [37].

In a holistic notion, according to the conceptual framework of DeSeCo, if competency is not reduced to its cognitive dimension, it has a broader meaning than a skill.

Key competencies shown in **Table 2**, as OECD remarks [37], are not determined by arbitrary decisions about what personal qualities and cognitive skills are desirable, but by a careful consideration of the factors required for a successful life and a well-functioning society.

Use tools interactively	Interact in heterogeneous groups	Act autonomously
Use communication skills effectively	Be empathetic	Understand patterns
Access adequate information sources	Manage emotions	Have an idea of the system
Evaluate the value of information	Present ideas and listen to others	Identify action consequences
Organize knowledge and information	Understand debate	Choose among available options
Use technology	Construct tactical alliances	Define projects and set goals
_	Negotiate	Evaluate necessary resources
_	Make decisions	Balance resources to meet goals
-	Analyze issues and interests	Learn from past actions
_	Identify areas of agreement	Monitor progress
_	Reframe the problem	Understand own interests
-	Prioritize needs and goals	Know rules and principles
-	_	Construct arguments
-	_	Suggest alternative solutions
Extract of [37].		

Table 2. Key competencies of DeSeCo project.

DeSeCo also sets a conceptual context for assessment by the Programme for International Student Assessment (PISA), considering their criticity according to three criteria:

- Contribution to highly valued outcomes at an individual and societal level
- Instrumentation for meeting important, complex demands and challenges
- Importance for all individuals, not just for specialists

In the Latin America, Portugal, and Spain contexts, only Chile, Colombia, Mexico, Spain, and Portugal are members of the OECD, being Brazil a partner and Costa Rica a guest. However, an OECD regional initiative covers the whole Latin American region.

In **Table 3**, OECD PISA 2015 results are shown [38]. It's observed a level of performance and resilience below the average, which contradicts with the levels of motivation, interest, and enjoyment.

3.2.2. Tuning project for university stage

Tuning project defines a competence, into an integrated approach, as the capability to execute the degree of preparation, sufficiency, and/or responsibility for certain tasks [39]. Also defines it as a dynamic combination of knowledge, understanding, skills, abilities and values [40] and as the capacities that all humans need to resolve the situations that arise in their lives effectively and autonomously [41]. To understand this concept of competence properly, it is necessary to include knowing how to:

- Understand: theoretical knowledge of an academic field
- · Act: practical and operational application of knowledge to certain situations
- Be: values as an integral element of the way of living in a social context

EHEA and ALFA Tuning projects propose a methodology [40, 41], whose framework is designed to understand and compare the curricula, based on these five approaches:

- 1. Generic competences
- 2. Specific competences
- 3. Learning, teaching, assessment, and performance
- 4. Role of European Credit Transfer System (ECTS) as an accumulation system
- 5. Role of quality enhancement in the educational process

Within this context, Tuning classifies generic competences, as Table 4 shows into three groups:

- Instrumental (cognitive abilities, methodological capacities, linguistic skills, and technological capabilities)
- Interpersonal (individual abilities and social skills)
- Systemic (abilities and skills concerning whole systems)

Rank		Performance	Efficacy	Enjoyment	Interest	Motivation	Resilience
_	OECD average	49%	24%	60%	53%	65%	29%
23	Portugal	50%	31%	73%	66%	72%	38%
30	Spain	49%	23%	61%	58%	68%	39%
38	Argentina	47%	31%	52%	_	66%	15%
44	Chile	44%	19%	67%	54%	68%	15%
47	Uruguay	43%	23%	64%	54%	72%	14%
53	Trinidad and Tobago	42%	37%	71%	_	78%	13%
55	Costa Rica	42%	24%	78%	60%	77%	9%
57	Colombia	41%	24%	79%	67%	78%	11%
58	Mexico	42%	27%	76%	70%	80%	13%
63	Brazil	40%	27%	77%	62%	81%	9%
64	Peru	39%	29%	79%	71%	85%	3%
70	Dominican Republic	34%	36%	84%	80%	82%	1%
Extra	ct of [38].						

Table 3. OECD PISA 2015 results.

Instrumental	Interpersonal	Systemic
Analysis and synthesis	Criticism and self-criticism	Applying knowledge in practice
Organization and planning	Teamwork	Research
Basic general knowledge	Interaction with technical experts	Learning
Communication	Working in heterogeneous teams	Adaptation to new situations
Elementary computing	Appreciation of diversity	Creativity
Information management	Working in international context	Leadership
Problem solving	Ethical commitment	Judgment of cultures and customs
Decision making	Motivation	Working autonomously
_	Cooperation	Project design and management
_	_	Initiative and entrepreneurial spirit
_	_	Concern for quality
_	_	Will to succeed

Table 4. EHEA and ALFA tuning project competences.

Tuning EHEA includes Portugal and Spain and Tuning ALFA counts with the participation of Argentina, Brazil, Bolivia, Colombia, Costa Rica, Cuba, Chile, Dominican Republic, Ecuador, El Salvador, Guatemala, Honduras, Mexico, Nicaragua, Panama, Paraguay, Peru, Uruguay, and Venezuela.

3.3. Soft skills in project management

Projects, as Jensen et al. affirm [42], have become omnipresent not only in economy but also in society. In fact, both DeSeCo in the preuniversity stage (defining projects and setting goals) and Tuning in the university stage (designing and managing projects) include projects in all students' curricula.

The importance of both hard skills (relating to processes) and soft skills (dealing with people) is widely recognized in project management (PM), as Azim et al. and Ahern et al. explain [43, 44], being managers, individually, responsible for balancing and optimizing their application. Chipulu et al., in the context of PM, extract six dimensions for them [45]:

- 1. Industry-specific and generic skills over project management knowledge/expertise
- 2. PM knowledge/expertise over industry-specific and generic skills
- 3. Managerial (senior) skills
- 4. Personal (positive) traits
- 5. Project management methodology experience and professional qualifications
- 6. Risk management over a project life cycle

For Ojiako et al. [46, 47], to learn properly about social and behavioral skills is critical in order to complete the role transformation from technicians to managers. Azim et al. advice that project managers and senior executives have to realize the significance of managing people [43]. In this context, soft skills are increasing their influence, being not only required but also important to manage teams. Pant and Baroudi identify soft skills as the missing link critical for achieving project success and advice the lack of emphasis placed on this within the context of university education [48]. Strang concludes that managing projects requires a series of skills, including interpersonal abilities, technical competencies, cognitive aptitudes, the capability to understand both context and people, and the integration of leadership behaviors [49]. Posner argues that rather than technical skills, intrapersonal and interpersonal competences are the most critical to solve problems in project management [50]. Likewise, for Chipulu et al., both managerial skills and personal traits are critical to manage complex environments characterized by rapid changes and uncertainty [45].

In summary, in the context of project management, there are a lot of studies that highlight the impact of soft skills on project success, as Strang [49], Pant and Baroudi [48], Awan et al. [51], López et al. [52], Kandelousi et al. [53], Cousillas et al. [54], Koutsikouri et al. [55], Camilleri [56], Davis [22], or Carmona-Chaves [57], among others, expose.

3.3.1. IPMA approach

IPMA is the World's first project management association (in particular, a federation of 68 national PM associations), founded in 1965, that offers unique, role-specific competence development guidelines, for improved project success.

IPMA competence baseline (ICB) is a global standard that defines the competences required by individuals working in the field of PM, to train (and certify) future professionals, who will probably work in distributed environments with overlapping and conflicting stakeholder interests, shaped by real-time data and performance management tools, challenged with too much information and not enough communication, and judged by their ability to deliver outcomes that align with short-and long-term strategies [58]. IPMA organizes the profession into three competence areas:

- People: Personal and interpersonal competences required to succeed in projects
- Practice: Technical aspects of managing projects
- Perspective: Contextual competences that must be navigated within environment

Table 5 shows people competences, as well as skills related to them.

3.3.2. PMI approach

PMI is the world's leading professional membership association for PM, founded in 1969, with over half a million members and certification holders in 185 countries. The guide to the project management body of knowledge (PMBOK) provides guidelines for managing individual projects and defines PM-related concepts, as methods, processes, and practices [59].

People competences	Skills related			
Communication	Facilitation, empathy, moderation, promotion			
Conflict and crisis	Creativity, moderation, persuasiveness, prevention			
Integrity and reliability	Confidence, consistency, equity, ethics, professionalism, responsibility, trustworthiness			
Leadership	Awareness, coaching, commitment, decision making, empowerment, flexibility, influence, learning, managing, mentoring, proactivity, team building			
Negotiation	Assertiveness, empathy, patience, persuasion, attitude			
Relations and engagement	Commitment, confidence, diversity, empathy, encouragement, engagement, facilitation, intuition, motivation, networking, resistance, team building			
Resourcefulness	Critical thinking, decision making, creativity, facilitation, innovation, problem solving, resilience			
Results orientation	Balance, delegation, effectiveness, efficiency, entrepreneurship, organization, productivity, sensitivity			
Self-reflection & self-management	Awareness, confidence, delegation, effectiveness, focusing on goals, motivation, organization, prioritization, relaxation, responsibility			
Teamwork	Cooperation, delegation, empowerment, facilitation, networking, organization, recruitment, time management, team building			
Extract of [58].				

Table 5. IPMA ICB People competences and skills related.

However, understanding and applying tools and techniques recognized as good practice are not enough to be effective. In addition to specific skills and general management proficiencies required for a project, it is necessary to domain the three competency dimensions, as PMI proposes in the project manager competency development framework (PMCDF) [60]:

- Knowledge: Knowing about PM
- Performance: Accomplishing while applying PM knowledge
- Personal: Behaving when performing the project or related activities

Table 6 Shows interpersonal skills (from PMBOK) and personal competences (from PMCDF).

3.4. Soft skills in workplace

Employability includes an array of technical and nontechnical skills, as well as knowledge, management, and experience, in order to ensure that a student is able to put them into practice, reason why they should be included into the educational stages, as Sangwan and Garg hold [61]. In the same way, Robles states that executives consider soft skills a very important attribute in job applicants [62]. According to Sutton, soft skills are extremely important for job hires in many occupations [63]. Besides, Truong et al. reveal that major employers value the potential role that soft skills can play in maximizing business success [64].
Interpersonal skills	Personal competences
Awareness	-
Coaching	_
_	Cognitive ability
Communication	Communicating
Conflict management	_
Decision making	_
_	Effectiveness
Influencing	_
Leadership	Leading
_	Managing
Motivation	_
Negotiation	_
_	Professionalism
Team building	_
Trust building	-
Extract of [59, 60].	

Table 6. PMI PMBOK and PMCDF interpersonal skills and personal competences.

For Mitchell et al. [28], in the twenty-first century, organizations seek versatile individuals, even for entry-level jobs. The integration of soft skills into students' curriculum also promotes their hiring in today's workforce, and their proficiency is important to potential employers. Lindsey and Rice recapitulate that successful graduates should possess a high ratio of emotional-social intelligence against book smarts [65]. In fact, as many authors as Saravanan [66], Ramlall and Ramlall [67] or Alismail and McGuire [68], among others, resume, employers are increasingly demanding a greater range of soft skills. In fact, as Robles holds [62], candidates, who add value with their soft skills, have the ability to make the difference in obtaining and retaining the jobs for which they have been prepared.

Pittenger et al. or Nealy consider soft skills a factor of equal importance to hard skills in career success [69, 15]. Other authors, as Wats and Wats or Klaus, claim that soft skills account for individual success more than hard skills [70, 27]. It is incontestable that soft skills play an integral role in success, as Bennett [71], Gibb [8], Schultz [72], Weber et al. [73], Sheikhy and Shafiee [74], Truong et al. [64], Tito Maya and Serrano Orellana [36], Holtzman and Kraft [75], or Nusrat [76], among others, remark.

From the studies published by Deloitte, EY, KPMG, and PwC, also known as the Big Four because they are the largest professional networks that offer their services in management consulting to majority of public and private companies all around the world, a compilation of the most demanded soft skills in the labor market is made.

Once the assemblage is collected, then only those that are repeated are transferred to **Table 7**, discarding other soft skills for employability, such as assertiveness, balance, coaching, commitment, compliance, confidence, conflict resolution, empowerment, encouragement, endurance, engagement, esteem, facilitation, honesty, illusionment, inclusiveness, judgment, monitoring, participation, perceptiveness, positive attitude, reasoning, recruitment, responsiveness, sociability, technology, and training.

Skills	Labor market demands				
	Deloitte	EY	KPMG	PwC	-
Adaptability	х		X	X	_
Appreciation		Х	Х		
Awareness	Х		Х		
Collaboration	Х			Х	
Communication	Х	Х	Х	Х	
Control		Х	Х		
Cooperation		Х	х		
Coordination	х			Х	
Creativity	Х			Х	
Critical thinking	х		х	х	
Customer service	х	Х		Х	
Decision making	Х		Х	Х	
Diversity		Х		х	
Ethics	х	Х	х		
Flexibility	Х	Х	Х		
Influence			х	х	
Initiative	Х		Х		
Innovation	Х			Х	
Integrity		Х		Х	
Leadership	Х	Х	Х	Х	
Learning	Х		Х		
Management	Х	Х	Х		
Mentoring	Х	Х			
Motivation			Х	Х	
Negotiation	Х	Х	х	Х	
Networking		Х	Х	Х	
Organization		Х	х		

Skills	Labor market d	lemands		
	Deloitte	EY	KPMG	PwC
Persuasion	Х		X	X
Planning		Х	х	
Problem solving	Х	Х	х	х
Professionalism	Х		х	
Sensitivity	Х		х	
Teamwork	Х		х	х
Time management			х	х
Trustworthiness		Х		х
Extract of [77–82].				

Table 7. Essential skills for labor market demands by Big Four.

4. Comparative analysis

Table 8 Summarizes the whole collection of soft skills from the literature review, but those skills that have only appeared on a single occasion have been eliminated, such as agility, appearance, authority, balance, citizenship, coaching, cooperation, coordination, directiveness, independence, investigation, loyalty, marketing, opportunity, patience, persuasion, proactivity, reliability, sensibility, and training.

5. Discussion

Comparative analysis from literature review made in **Table 8** has to be contrasted against Big Four's compilation. Then, once the most demanding skills have been selected, it is necessary to check if PM frameworks chosen incorporate them, in order to establish a model of practical implementation during the formative stages, especially the university.

Firstly, there are soft skills that are highlighted almost unanimously: communication, teamwork, problem solving and/or conflict resolution, critical thinking, self-reflexion and selfmanagement, and leadership.

Secondly, other soft skills are emphasized by the majority of authors, educators, trainers, practitioners, consultants, and employers: ethics and/or integrity, creativity, active learning, motivation, attention to diversity, and professionalism and/or reliability.

Thirdly, it is convenient to accentuate other soft skills that are very present among consultants, PM frameworks, and educational projects, which have hardly been studied, in general, by authors and researchers: decision making, result orientation, influence and/or persuasion, coaching and mentoring, negotiation, and coordination and/or cooperation.

Skills	Educ	cation	ial stag	ges					Projec	ct man:	agemei	ıt				Worl	cplace					
	[30]	[9]	[72]	[17]	[32]	[34]	[21]	[64]	[50]	[83]	[43]	19] [4	[9]	l] [75]	Ξ	5	[73]	[28]	18] [3	1] [67]	[62]	[92]
Adaptability												×	×			×						
Awareness	×	×	×				×			×		~			×				×			
Collaboration				×						×		~			×							
Commitment		×					×									×	×					
Communication	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×		×	×	×	×	×
Confidence		×			×		×						×									
Courtesy			×													×	×			×	×	
Creativity			×	×		×			×	×		~			×			×		×		
Critical thinking		×	×	×		×		×		×		×		×	×			×	×			×
Customer service		×			×											×		×				
Decision making						×				×		~							×			
Delegation									×		×		×									
Development		×											×									
Diversity	×		×			×								×			×	×	×			
Empathy			×					×	×													
Entrepreneurship			×													×						
Enthusiasm													×	×		×	×					
Ethics			×		×			×						×	×			×	×	×	×	
Flexibility	×	×						×	×				×	×	×					×	×	
Honesty			×													×	×		×			
Influence		×															×					
Initiative	×	×		×												×						

Skills	Educ	ation	al stag	sa					Projec	t mané	Igemei	nt				M	rkplace						
	[30]	[9]	[72]	[17]	[32]	[34]	[21]	[64]	[50]	[83]	[43]	[19]	46] [5	2] [E	5] [1]	[7]	[73]	[28]	[18]	[31]	[67]	[62]	[92]
Innovation				×						×		×				×			×				
Integrity			×		×											×	×			×		×	
Leadership		×		×	×	×	×	×			×		×	×				×	×		×		×
Learning		×							×	×		×	~		×	×				×			
Motivation					×			·	×		×			×		×	×			×			×
Negotiation			×					×															
Networking					×			×						×									
Planning	×	×	×					·	×				~	×						×			
Positive attitude								×												×	×	×	
Problem solving			×	×		×	×	×		×		×	×		×	×		×	×	×			×
Professionalism								×						×			×		×	×	×	×	
Responsibility			×	×				×		×	×	×							×	×	×	×	
Result orientation		×					×		×														×
Self-organization	×	×	×			×	×	×	×			~	~	×	×	×		×		×			×
Socialization			×				×			×		×	×	×			×		×	×	×	×	×
Teamwork	×	×	×			×	×	×	×	×	×	×	×	×		×		×	×	×	×	×	×
Technology	×	×							-	×		\sim	>		х	×			x				

, and workplace.
ΡM
education,
Е.
review
literature
from
skills
century
wenty-first
Ĥ.
e œ
Tabl

Later, it is opportune to rescue other soft skills whose interest grows over time, finding more and more references in recent literature: competitiveness, customer service, initiative and/or entrepreneurship, marketing and publicity, and sustainability.

To conclude, from the compilation of the soft skills developed, it is opportune to discuss how these skills can be grouped, for which how the authors have organized them is analyzed.

Cheng, Dainty and Moore, and Le Deist and Winterton, in the context of human resources management, classify competences into three groups [84, 85]:

- Functional (job-specific skills)
- Cognitive (knowledge and understanding)
- Social (behavioral and attitudinal)

In the same vein, Binkley et al. organize twenty-first century skills into three groups [83]:

- Ways of thinking
- Ways and tools for working
- Living in the world

Similarly, Onisk classifies generic soft skills into three broad categories [86]:

- Behavioral development: improving (or enhancing) the underlying social behaviors and influencing capabilities
- Professional development: obtaining (or maintaining) a professional certification or accreditation
- Compliance: helping employers become legally compliant with legislated standards

According to classifications studied, selected skills are organized in Table 9.

Intrapersonal		Interpersonal
Cognitive	Individual	Social
Active learning	Competitiveness	Attention to diversity
Communication	Empathy and/or sensitiveness	Coordination and/or cooperation
Creativity	Ethics and/or integrity	Customer service
Critical thinking	Initiative and/or entrepreneurship	Influence and/or persuasiveness
Decision making	Leadership	Marketing and publicity
Problem solving and/or conflict resolution	Motivation	Mentoring and/or training
Results orientation	Professionalism and/or reliability	Negotiation
Self-reflection and self-management	Sustainability	Teamwork

Table 9. Traceability of intra- and interpersonal skills from education to labor market.

6. Conclusions

Despite there are different definitions for intra- and interpersonal skills' set, a general consensus about its main characteristics can be made: they are social abilities, individual attributes, and cognitive attitudes. They are also called as twenty-first century skills, essential skills, human skills, professional skills, and/or soft skills.

Intra- and interpersonal skills are currently essential for both individual careers and organizational success, being identified by many employers as the number one differentiator, regardless of the type of organization. Importance of intra- and interpersonal skills has increased more and more over last years among new employees.

The gap between skill levels that employers need from recent graduates and new employees versus skill development grade they own is growing. To decrease the contrast between supply and demand, the implementation of training programs throughout all educational stages is required, from high school to university.

Into a knowledge-based economy, with abundant unskilled human resources, it is time to train recent graduates with properly employable skills. It is necessary to emphasize that education should focus not only on core academic subject mastery, but also on intra- and interpersonal skills development.

Labor market demands that employees acquire and/or improve their intra- and interpersonal skills around three dimensions:

- Cognitive skills: being creative, communicating, focusing on results, learning actively, making decisions, resolving conflicts, solving problems, self-managing and self-reflecting, and thinking critically
- Individual skills: being competitive, entrepreneurial, ethic, integrous, professional, reliable and sensitive, empathizing, leading, motivating, promoting sustainability, and taking the initiative
- Social skills: attending to diversity, being persuasive, coordinating, cooperating, influencing, marketing, mentoring, negotiating serving customers, training, and working as a team

PM competencies approach can be used to develop and perfect intra- and interpersonal skills of employees, through formation programs that train how to be aware culturally and politically, be effective, integrous, professional, related, reliable and resourceful, build trust, coach, communicate, engage, influence, lead, make decisions, manage, motivate, negotiate, orientate to results, resolve conflicts, and work as a team.

Training practices of PM by competencies can be introduced in education. In fact, PM proposals are included on theoretical educational frameworks exposed (defining, designing and managing projects and setting goals, besides balancing, executing, evaluating, interacting, monitoring, organizing, planning, prioritizing, among other actions intrinsically related to leading projects). In the context of Portugal, Spain, and Latin America countries, the theoretical frameworks that cover the different educational stages include most of the soft skills selected:

- During the preuniversity stage, DeSeCo Project by OECD tries to instill that students assert rights and duties, communicate, conduct plans and projects, construct alliances, cooperate, empathize, make decisions, negotiate, recognize merits, resolve conflicts, be self-aware, suggest alternatives, support others, and take responsibility, among other skills
- At the university stage, EHEA and ALFA Tuning Projects ensure that future graduates analyze, appreciate diversity, are competitive, be creative and critical, commit, communicate, lead, learn, make decisions, motivate, solve problems, synthesize, take initiative, and work as a team, among other skills

7. Future research

Once it has been proven that, from a theoretical point of view, most of the soft skills demanded by the labor market are present in education (including to direct and manage projects), it is mandatory to ask right now what is wrong. There are two nonexclusive possibilities:

• Learning and training of soft skills is not done or done weakly

(subordinating them to hard skills)

• Assessment of soft skills' performance is not done or done incorrectly

(avoiding their application's improvement)

To answers these questions, further research is needed:

- Launching a questionnaire in Portugal, Spain, and Latin American countries, directed to recent graduates and new employees, to check the importance given by the system to soft skills they have perceived during their education and contrast the importance they think soft skills are going to reach, in order to measure the gap between educational criticity and workplace impact
- Analyzing assessment methods and checking if they are adequate and effective, proposing improvements

Author details

Alberto Cerezo-Narváez1*, María José Bastante Ceca2 and José Luis Yagüe Blanco3

*Address all correspondence to: alberto.cerezo@uca.es

- 1 Universidad de Cádiz UCA, Spain
- 2 Universitat Politècnica de València UPV, Spain
- 3 Universidad Politécnica de Madrid UPM, Spain

References

- van Laar E, van Deursen AJAM, van Dijk JAGM, et al. The relation between 21st century skills and digital skills: A systematic literature review. Computers in Human Behavior. 2017;**72**:577-588. DOI: 10.1016/j.chb.2017.03.010
- [2] Sliter KA. Assessing 21st century skills: Competency Modeling to the rescue. Industrial and Organizational Psychology: Perspectives on Science and Practice. 2015;8:284-290. DOI: 10.1017/iop.2015.35
- [3] Neubert JC, Mainert J, Kretzschmar A, et al. The assessment of 21st century skills in industrial and organizational psychology: Complex and collaborative problem solving. Industrial and Organizational Psychology: Perspectives on Science and Practice. 2015;8:238-268. DOI: 10.1017/iop.2015.14
- [4] Rodriguez D, Patel R, Bright A, et al. Developing competency models to promote integrated human resource practices. Human Resource Management. 2002;41:309-324. DOI: 10.1002/hrm.10043
- [5] Bonilla Climent JB. La Educación Basada en Competencias como Instrumento de Política Educativa y Laboral. Revista Mexicana de Agronegocios. 2008;**XII**(22):490-502
- [6] Rainsbury E, Hodges D, Burchell N, et al. Ranking workplace competencies: Student and graduate perceptions. Asia-Pacific Journal of Cooperative Education. 2002;**3**(2):8-18
- [7] Cleary M, Flynn R. Thomasson S. Employability Skills From Framework to Practice. Melbourne: Commonwealth of Australia; 2006
- [8] Gibb J. Generic Skills and Training. Adelaide: National Centre for Vocational Education Research; 2004
- [9] Dictionaries C. Collins English Dictionary. 11th ed. Glasgow: HarperCollins; 2011
- [10] Dictionaries O. Oxford Dictionary of English. 3rd ed. Oxford: Oxford University Press; 2010
- [11] Perreault HR. Business educators can take a leadership role in character education. Business Education Forum. 2004;59:23-25
- [12] James RF, James ML. Teaching career and technical skills in a 'mini' business world. Business Education Forum. 2004;**59**(2):39-41
- [13] Su R, Golubovich J, Robbins SB. Bridging science and practice: Toward a standard, evidence-based framework of 21st century skills. Industrial and Organizational Psychology: Perspectives on Science and Practice. 2015;8:289-295. DOI: 10.1017/iop.2015.36
- [14] Kraiger K. Perspectives on Training and Development. In: Borman WC, Ilgen DR, Klimoski RJ, et al., editors. Handbook of psychology. Vol 12: Industrial and Organizational Psychology. New Jersey: John Wiley & Sons; 2003. p. 171-192. DOI: 10.1002/0471264385.wei1208
- [15] Nealy C. Integrating soft skills through active learning in the management classroom. Journal of College Teaching & Learning. 2005;2:1-6. DOI: 10.19030/tlc.v2i4.1805

- [16] Gewertz C. 'Soft Skills' in Big Demand. Education Week. 2007;26:25-27
- [17] Ziegenfuss RM. Education in the 21st Century: Toward an Expanded Epistemic Frame of Leadership [thesis]. Philadelphia: University of Pennsylvania; 2010
- [18] Hodge KA, Lear JL. Employment skills for 21st century workplace: The gap between faculty and student perceptions. Journal of Career and Technical Education. 2011;26:28-41
- [19] Suto I. 21st Century skills: Ancient, ubiquitous, enigmatic?. Research Matters: A Cambridge Assessment Publication. 2013;15:2-8
- [20] Soland J, Hamilton LS, Stecher BM. Measuring 21st Century Competencies. Santa Mónica: Guidance for Educators; 2013
- [21] Hayes JH. Skill Builders: Perceived Skills Enhanced by Students through Participation in High School Extracurricular Activities [thesis]. Boiling Springs: Gardner-Webb University; 2014
- [22] Davis K. Different stakeholder groups and their perceptions of project success. International Journal of Project Management. 2014;32:189-201. DOI: 10.1016/j.ijproman. 2013.02.006
- [23] Schooner P, Nordlöf C, Klasander C, et al. Developing 21st century skills in Swedish compulsory school technology education: Three teacher perspectives. In: 32 PATT Conference: Technology Education for 21st Century Skills. Utrecht: Delft University of Technology and HU University of Applied Sciences; 2016. p. 411-418
- [24] Ali SM, Harun H, Massari N, et al. The 21st century skills in online multiliteracies project approach (eMULPA): Learners' reflections on their knowledge processes. Mediterranean Journal of Social Sciences. 2017;8:252-258. DOI: 10.5901/mjss.2017.v8n1p252
- [25] Cullen C. El debate epistemológico de fin de siglo y su incidencia en la determinación de las competencias científico tecnológicas en los diferentes niveles de la educación formal. Parte II. Novedades Educativas. 1996;62:1-20
- [26] Bronson E. Helping CTE students learn to their potential. Techniques: Connecting Education and Careers. 2007;82:30-31
- [27] Klaus P. Communication Breakdown. California Job Journal. 2010;28:1-9
- [28] Mitchell GW, Skinner LB, White BJ. Essential soft skills for success in the twenty first century workforce as perceived by business educators. Delta Pi Epsilon Journal. 2010;51:43-53
- [29] Tan AYT, Chew E, Kalavally V. The expectations gap for engineering field in Malaysia in the 21st century. On the Horizon. 2017;25:131-138. DOI: 10.1108/OTH-12-2015-0071
- [30] Nabi GR, Bagley D. Graduates' perceptions of transferable personal skills and future career preparation in the UK. Career Development International. 1998;3:31-39. DOI: 10.1108/13620439810368619

- [31] International Youth Foundation. Getting Youth in the Door: Defining Soft Skills Requirements for Entry-level Service Sector Jobs. Baltimore: International Youth Foundation; 2013
- [32] Laud R, Johnson M. The future of the MBA curriculum : Improving relevancy through evidenced- based soft skills. In: Academy of Educational Leadership. New Orleans: Allied Academies International Conference; 2012. p. 63-67
- [33] Hassan A, Maharoff M, Abiddin NZ. The Readiness of Lecturers in Embedding Soft Skills in the Bachelor's Degree Program in Malaysian Institutes of Teacher Education. 2014;2:138-143. DOI: 10.11114/jets.v2i3.455
- [34] Low M, Samkin G, Liu C. Accounting Education and the Provision of Soft Skills: Implications of the recent NZICA CA Academic requirement changes. e-Journal of Business Education & Scholarship of Teaching. 2013;7(1):1-33
- [35] Larraz N, Vázquez S, Liesa M. Transversal skills development through cooperative learning. Training teachers for the future. On the Horizon. 2017;25:85-95. DOI: 10.1108/ OTH-02-2016-0004
- [36] Tito Maya MD, Serrano OB. Desarrollo de soft skills: una alternativa a la escasez de talento humano. INNOVA Research Journal. 2016;1:59-76
- [37] Organisation for Economic Co-operation and Development. The definition and selection of key competencies - Executive summary. Paris: OECD Publishing; 2005. DOI: 10.1080/ 2159676X.2012.712997
- [38] Organisation for Economic Co-operation and Development. PISA. Results. Excellence and Equity in Education. Paris: OECD Publishing; 2015. p. 2015
- [39] González J, Wagenaar R. Tuning educational structures in Europe. In: Pilot Project -Phase. Vol. 1. Publicaciones de la Universidad de Deusto: Bilbao; 2003
- [40] González J, Wagenaar R. Tuning Educational Structures in Europe. Universities' Contribution to the Bologna Process. Publicaciones de la Universidad de Deusto: Bilbao; 2005
- [41] Beneitone P, Esquetini C, González J, et al. Latin America Project. Tuning Reflections on and Outlook for Higher Education in Latin America. Publicaciones de la Universidad de Deusto: Bilbao; 2007
- [42] Jensen AF, Thuesen C, Geraldi J. The projectification of everything: Projects as a human condition. Project Management Journal. 2016;47:21-34
- [43] Azim S, Gale A, Lawlor-Wright T, et al. The importance of soft skills in complex projects. International Journal of Managing Projects in Business. 2010;3:387-401. DOI: 10.1108/ 17538371011056048
- [44] Ahern T, Leavy B, Byrne PJ. Knowledge formation and learning in the management of projects: A problem solving perspective. International Journal of Project Management. 2014;32:1423-1431. DOI: 10.1016/j.ijproman.2014.02.004

- [45] Chipulu M, Neoh JG, Ojiako U, et al. A multidimensional analysis of project manager competences. IEEE Transactions on Engineering Management. 2013;60:506-517. DOI: 10.1109/TEM.2012.2215330
- [46] Ojiako U, Chipulu M, Ashleigh M, et al. Project management learning: Key dimensions and saliency from student experiences. International Journal of Project Management. 2014;32:1445-1458. DOI: 10.1016/j.ijproman.2014.02.002
- [47] Ojiako U, Chipulu M, Marshall A, et al. Project Management learning: A comparative study between engineering students' experiences in South Africa and the United Kingdom. Project Management Journal. 2015;46:47-62. DOI: 10.1002/pmj.21510
- [48] Pant I, Baroudi B. Project management education: The human skills imperative. International Journal of Project Management. 2008;26:124-128. DOI: 10.1016/j.ijproman.2007.05.010
- [49] Strang K. Achieving Organizational Learning across Projects. In: PMI® Global Congress 2003 - North America. Baltimore: PMI; 2003. p. 10
- [50] Posner BZ. What it takes to be a good project manager. Project Management Journal. 1987;18:51-54
- [51] Awan MH, Ahmed K, Zulqarnain W. Impact of project Manager's soft leadership skills on project success. Journal of Poverty, Investment and. Development. 2015;8:27-47
- [52] López Paredes A, Pajares Gutierrez J, Iglesias Sanzo M. Certificación IPMA-4LC. Manual de Preparación. In: Valladolid: Business Project Management Solutions & Technologies. 2013
- [53] Kandelousi NS, Ooi J, Abdollahi A. Key success factors for managing projects. International Scholarly and Scientific Research & Innovation. 2011;5:1185-1189
- [54] Cousillas SM, Rodríguez-Montequín V, Villanueva-Balsera J, et al. Análisis de factores de éxito y causas de fracaso en proyectos: Herramientas de patrones de comportamiento mediante técnicas. In: XVII Congreso Internacional de Dirección e Ingeniería de Proyectos. Logroño: AEIPRO. 2013. p. 190-202
- [55] Koutsikouri D, Dainty A, Austin S. Critical success factors for multidisciplinary engineering projects. In: 22nd Annual ARCOM Conference. Birmingham: Association of Researchers in Construction Management; 2006. p. 219-228
- [56] Camilleri E. Project Sucess: Critical Factors and Behavoiurs. Burlington: Gower Publishing Company; 2011
- [57] Carmona-Chaves A-I. Las diez principales competencias de un lider de proyectos. In: San José. 2013
- [58] International Project Management Association. Individual Competence Baseline for Project, Programme & Portfolio Management. 4th ed. Zurich: IPMA; 2015
- [59] Project Management Institute. A Guide To. The Project Management Body of Knowledge. PMBOK Guide - Fifth Edition. 5th ed. Newtown Square: PMI; 2013

- [60] Project Management Institute. Project Manager Competency Development Framework. 3rd ed. Newtown Square: PMI; 2017
- [61] Sangwan S, Garg S. WIL and business graduate skill transfer to workplace. On the Horizon. 2017;25:109-114. DOI: 10.1108/OTH-06-2016-0031
- [62] Robles MM. Executive perceptions of the top 10 soft skills needed in Today's workplace. Business and Professional Communication Quarterly. 2015;75:453-465. DOI: 10.1177/1080569912460400
- [63] Sutton N. Why can't we all just get along?. Computing Canada. 2002;28(16):1-20
- [64] Truong HTT, Laura RS, Shaw K. New insights for soft skills development in Vietnamese business schools: Defining essential soft skills for maximizing graduates' career success. International Journal of Social, Behavioral, Educational, Economic, Business and. Industrial Engineering. 2016;10:1857-1863
- [65] Lindsey NS, Rice ML. Interpersonal skills and education in the traditional and online classroom environments. Journal of Interactive Online Learning. 2015;**13**:126-136
- [66] Saravanan V. Sustainable employability skills for engineering professionals. The Indian Review of World Literature in English. 2009;5:1-9
- [67] Ramlall S, Ramlall D. The value of soft-skills in the accounting profession: Perspectives of current accounting students. Advances in Research. 2014;2:645-654. DOI: 10.9734/AIR/2014/11000
- [68] Alismail HA, Mcguire P. 21st century standards and curriculum: Current research and practice. Journal of Education and. Practice. 2015;6:150-155
- [69] Pittenger KKS, Miller MC, Mott J. Using real-world standards to enhance students' presentation skills. Business and Professional Communication Quarterly. 2004;67:327-336. DOI: 10.1177/1080569904268084
- [70] Wats M, Wats RK. Developing soft skills in students. International Journal of Learning. 2009;15:1-10. DOI: 10.18848/1447-9494/CGP/v15i12/46032
- [71] Bennett R. Employers' demands for personal transferable skills in graduates: A content analysis of 1000 job advertisements and an associated empirical study. Journal of Vocational Education & Training. 2002;54:457-476. DOI: 10.1080/13636820200200209
- [72] Schulz B. The importance of soft skills: Education beyond academic knowledge. Journal of. Language & Communication. 2008;2:19-29
- [73] Weber MR, Finley DA, Crawford A, et al. An exploratory study identifying soft skill competencies in entry-level managers. Tourism and Hospitality Research. 2009;9:353-361. DOI: 10.1057/thr.2009.22
- [74] Sheikhy A, Shafiee A. An investigation into the association between personal skills, organizational learning, and innovation, and organizational performance in the power distribution company, northern Kerman. Mediterranean Journal of Social Sciences. 2015;6:715-721. DOI: 10.5901/mjss.2015.v6n6s2p715

- [75] Holtzman DM, Kraft EM. Skills needed in the workplace: A comparison of the results of feedback from representatives of large and small businesses in New Jersey. International Journal of Management and Marketing Research. 2016;9:13-24
- [76] Nusrat M. Soft Skills for Sustainable Employment; Does it Really Matter? Dhaka; 2016
- [77] Knowles-Cutler A, Lewis H. Talent for survival. London: Essential skills for humans working in the machine age; 2016
- [78] Brooke B, Pyron DA, Matthews P, et al. Paradigm shift. Building a new talent management model to boost growth. London: Ernst & Young; 2012
- [79] KPMG's Business Academy. Leadership and business skills. Amstelveen: KPMG International; 2016
- [80] KPMG's Business Academy. Soft skills Courses & Workshops. Amstelveen: KPMG International; 2016
- [81] Bruce A. People: Growing Headcount and Access to Key Skills. London: Pricewater house Coopers; 2017
- [82] PwC's Academy. The Talent Challenge: Harnessing the Power of Human Skills in the Machine Age. New York: Pricewater house Coopers; 2017
- [83] Binkley M, Erstad O, Herman J, et al. Defining 21st century skills. Draft white paper 1. Melbourne: The University of Melbourne; 2010
- [84] Cheng MI, Dainty ARJ, Moore DR. What makes a good project manager? Human Resource Management Journal. 2005;15:25-37. DOI: 10.1111/j.1748-8583.2005.tb00138.x
- [85] Le Deist FD, Winterton J. What is competence? Human resource development international. 2005;8:27-46. DOI: 10.1080/1367886042000338227
- [86] Onisk M. Is Measuring Soft-Skills Training Really Possible? Sidney: Appcon; 2011



Edited by Manuel Otero-Mateo and Andres Pastor-Fernandez

People's competence has a strong influence on the strategy of human resource management, affecting daily aspects, thought patterns, and behavioral modes of executive management and employees. From a business perspective, there is a strong relationship between human capital and success and also an integral development of the human factor in all its dimensions, both personal and professional, and social competences must be a key factor to reach it. To help achieve this business excellence, it is necessary to transfer the demands of the labor market into education, and one of the ways is through methodological framework for Project Management and Management, specifically the IPMA Individual Competence Baseline, which is an essential tool for achieving economic growth, corporate development, and competitiveness.







